

Citation: Lambert AM, Parretti HM, Pearce E, Price MJ, Riley M, Ryan R, et al. (2022) Temporal trends in associations between severe mental illness and risk of cardiovascular disease: A systematic review and meta-analysis. PLoS Med 19(4): e1003960. https://doi.org/10.1371/journal.pmed.1003960

Academic Editor: Alex Dregan, King's College London, UNITED KINGDOM

Received: July 12, 2021

Accepted: March 8, 2022

Published: April 19, 2022

Copyright: © 2022 Lambert et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This report presents independent research funded by the National Institute for Health and Care Research (NIHR). AML and TM are supported by the NIHR Applied Research Collaboration (ARC) West Midlands. HMP (NIHR Academic Clinical Lectureship) was funded by the NIHR during some of this research. MJP was supported by the NIHR Birmingham Biomedical RESEARCH ARTICLE

Temporal trends in associations between severe mental illness and risk of cardiovascular disease: A systematic review and meta-analysis

Amanda M Lambert^{1*}, Helen M Parretti², Emma Pearce¹, Malcolm J Price^{1,3}, Mark Riley¹, Ronan Ryan¹, Natalie Tyldesley-Marshall¹, Tuba Saygın Avşar⁴, Gemma Matthewman¹, Alexandra Lee¹, Khaled Ahmed¹, Maria Lisa Odland^{1,5,6,7}, Christoph U. Correll^{8,9,10}, Marco Solmi^{11,12,13}, Tom Marshall¹

1 Institute of Applied Health Research, University of Birmingham, Birmingham, United Kingdom, 2 Norwich Medical School, University of East Anglia, Norwich, United Kingdom, 3 NIHR Birmingham Biomedical Research Centre, University Hospitals Birmingham NHS Foundation Trust and University of Birmingham, Birmingham, United Kingdom, 4 Department of Applied Health Research, University College London, London, United Kingdom, 5 Department of Obstetrics and Gynecology, St Olavs Hospital, Trondheim University Hospital, Trondheim, Norway, 6 Malawi-Liverpool-Wellcome Trust Research Institute, Blantyre, Malawi, 7 Institute of Life Course and Medical Sciences, University of Liverpool, United Kingdom, 8 The Zucker Hillside Hospital, Department of Psychiatry, Northwell Health, Glen Oaks, New York, United States of America, 9 Donald and Barbara Zucker School of Medicine at Hofstra/Northwell, Department of Psychiatry and Molecular Medicine, Hempstead, New York, United States of America, 10 Charité Universitäsmedizin Berlin, Department of Child and Adolescent Psychiatry, Berlin, Germany, 11 Department of Psychiatry, University of Ottawa, Ontario, Canada, 12 Department of Mental Health, The Ottawa Hospital, Ontario, Canada, 13 Ottawa Hospital Research Institute (OHRI), Clinical Epidemiology Program, University of Ottawa, Ontario, Canada

* AML759@student.bham.ac.uk

Abstract

Background

Severe mental illness (SMI; schizophrenia, bipolar disorders (BDs), and other nonorganic psychoses) is associated with increased risk of cardiovascular disease (CVD) and CVD-related mortality. To date, no systematic review has investigated changes in population level CVD-related mortality over calendar time. It is unclear if this relationship has changed over time in higher-income countries with changing treatments.

Methods and findings

To address this gap, a systematic review was conducted, to assess the association between SMI and CVD including temporal change. Seven databases were searched (last: November 30, 2021) for cohort or case–control studies lasting \geq 1 year, comparing frequency of CVD mortality or incidence in high-income countries between people with versus without SMI. No language restrictions were applied. Random effects meta-analyses were conducted to compute pooled hazard ratios (HRs) and rate ratios, pooled standardised mortality ratios (SMRs), pooled odds ratios (ORs), and pooled risk ratios (RRs) of CVD in those with versus without SMI. Temporal trends were explored by decade. Subgroup analyses by age, sex,

Research Centre at the University Hospitals Birmingham NHS Foundation Trust and the University of Birmingham. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: CUC has been a consultant and/or advisor to or has received honoraria from: AbbVie, Acadia, Alkermes, Allergan, Angelini, Aristo, Axsome, Damitsa, Gedeon Richter, Hikma, IntraCellular Therapies, Janssen/J&J, Karuna, LB Pharma, Lundbeck, MedAvante-ProPhase, MedInCell, Medscape, Merck. Mitsubishi Tanabe Pharma. Mylan. Neurocrine, Noven, Otsuka, Pfizer, Recordati, Rovi, Servier, Sumitomo Dainippon, Sunovion, Supernus, Takeda, Teva, and Viatris. He provided expert testimony for Janssen and Otsuka. He served on a Data Safety Monitoring Board for Lundbeck, Rovi, Supernus, and Teva. He has received grant support from Janssen and Takeda. He is also a stock option holder of LB Pharma. MS received honoraria/has been a consultant for Angelini, Lundbeck, Otsuka. MLO receives funding from The Liaison Committee for education, research and innovation in Central Norway.

Abbreviations: BD, bipolar disorder; CHD, coronary heart disease; CI, confidence interval; CVA, cerebrovascular accident; CVD, cardiovascular disease; DALY, disability-adjusted life year; DSM, Diagnostic and Statistical Manual of Mental Disorders; FGA, first-generation antipsychotic; HR, hazard ratio; ICD, International Classification of Diseases; MOOSE, Meta-analysis Of Observational Studies in Epidemiology; NOS, Newcastle-Ottawa scale; OR, odds ratio; PRISMA, Preferred Reporting Items for Systematic Review and Meta-Analyses; RR, risk ratio; SGA, secondgeneration antipsychotic; SIR, standardised incidence ratio; SMI, severe mental illness; SMR, standardised mortality ratio; WHO, World Health Organization.

setting, world region, and study quality (Newcastle-Ottawa scale (NOS) score) were conducted. The narrative synthesis included 108 studies, and the quantitative synthesis 59 mortality studies (with (\geq 1,841,356 cases and 29,321,409 controls) and 28 incidence studies (>401.909 cases and 14.372,146 controls). The risk of CVD-related mortality for people with SMI was higher than controls across most comparisons, except for total CVD-related mortality for BD and cerebrovascular accident (CVA) for mixed SMI. Estimated risks were larger for schizophrenia than BD. Pooled results ranged from SMR = 1.55 (95% confidence interval (CI): 1.33 to 1.81, p < 0.001), for CVA in people with BD to HR/rate ratio = 2.40 (95%) CI: 2.25 to 2.55, p < 0.001) for CVA in schizophrenia. For schizophrenia and BD, SMRs and pooled HRs/rate ratios for CHD and CVD mortality were larger in studies with outcomes occurring during the 1990s and 2000s than earlier decades (1980s: SMR = 1.14, 95% CI: 0.57 to 2.30, p = 0.71; 2000s: SMR = 2.59, 95% CI: 1.93 to 3.47, p < 0.001 for schizophrenia and CHD) and in studies including people with younger age. The incidence of CVA, CVD events, and heart failure in SMI was higher than controls. Estimated risks for schizophrenia ranged from HR/rate ratio 1.25 (95% CI: 1.04 to 1.51, p = 0.016) for total CVD events to rate ratio 3.82 (95% CI: 3.1 to 4.71, p < 0.001) for heart failure. Incidence of CHD was higher in BD versus controls. However, for schizophrenia, CHD was elevated in higher-quality studies only. The HR/rate ratios for CVA and CHD were larger in studies with outcomes occurring after the 1990s. Study limitations include the high risk of bias of some studies as they drew a comparison cohort from general population rates and the fact that it was difficult to exclude studies that had overlapping populations, although attempts were made to minimise this.

Conclusions

In this study, we found that SMI was associated with an approximate doubling in the rate ratio of CVD-related mortality, particularly since the 1990s, and in younger groups. SMI was also associated with increased incidence of CVA and CHD relative to control participants since the 1990s. More research is needed to clarify the association between SMI and CHD and ways to mitigate this risk.

Author summary

Why was this study done?

- Previous research has identified higher incidence and mortality from cardiovascular disease (CVD) in people with severe mental illness (SMI) than those without SMI.
- It is not known whether the gap has changed over time.

What did the researchers do and find?

• We conducted a comprehensive systematic review and meta-analysis of 108 cohort and case–control studies, including over 30 million participants.

- Compared to the general population, people with SMI, particularly schizophrenia, generally had higher levels of cardiovascular-related mortality and a higher incidence of CVD.
- For schizophrenia and bipolar disorder (BD), the rate ratios for coronary heart disease (CHD) mortality and cardiovascular mortality were larger in the 1990s and 2000s than in earlier studies, and the rate ratios were larger in younger people.

What do these findings mean?

- We confirmed a strong association between SMI and incidence of and mortality from CVD.
- This association became stronger in the 1990s and 2000s.

Introduction

Mortality in people with mental disorders is more than 2.2 times that of the general population and more than 2.5 times in those with psychoses, according to a systematic review and metaanalysis of 148 studies [1]. A recent estimate suggested that mental health disorders could be associated with 14.3% of all deaths across the globe, corresponding to around 8 million deaths each year [1].

The increased premature mortality seems to be the highest in schizophrenia spectrum disorders (schizophrenia), who die around 14.5 years earlier than people without schizophrenia, according to data from 11 studies and 247,603 patients [2]. In a large national cohort study, the mortality rate ratio was 3.2 in 104,000 people with schizophrenia [3].

In bipolar disorder (BD), figures are also concerning, with life expectancy decreased by 9 to 12 years in males and 8 to 10.5 years in females [4]. In a large cohort study, the mortality rate ratio was around 2.4 in 58,000 participants with BD compared with the general population [3].

More than two-thirds of the increased mortality in people with mental health disorders has been estimated to be due to natural causes [1]. When looking at cause-specific mortality in those with mental health disorders, cardiovascular disease (CVD) plays a major role [5]. According to a recent large national cohort study in Taiwan, the standardised mortality ratio (SMR) for CVD-related mortality was 3.02 (95% confidence interval (CI): 2.80 to 3.24) in those with schizophrenia and 1.87 (95% CI: 1.70 to 2.05) for BD [3].

While excess mortality from CVD for people with severe mental illness (SMI) is established, the relationship with CVD incidence is less clear. CVD is associated with modifiable risk factors such as smoking, hypertension, obesity, dyslipidemia, diabetes, alcohol consumption, and physical inactivity. People with SMI are known to have higher prevalence of risk factors that are associated with CVD [6] and face barriers to healthcare that reduce their ability to benefit from effective treatment [7–11]. Despite high CVD-related mortality rates, CVD may remain undiagnosed in people with schizophrenia and BD. A recent study found that people with schizophrenia were 66% less likely to be diagnosed with CVD prior to CVD-related death than individuals without SMI [12]. Women with BD were 38% less likely than control participants to be diagnosed.

The disease burden of both SMI and CVD is greater among higher-income countries compared with middle- or lower-income countries. Mortality from CVD is particularly high among high-income countries (294 deaths per 100,000 compared with 239.9 globally) [13]. There is also variation in the prevalence of schizophrenia, with a higher prevalence in developed countries compared to less developed countries (median 3.3 versus 2.6 per 1,000) [14]. The course, severity, and outcome of schizophrenia are generally worse in high-income compared with low-income countries. For example, in a World Health Organization (WHO) 10-country study, remission of their condition was experienced by 62.7% of patients with schizophrenia in developing countries, compared with only 36.8% in developed countries [15]. The rates of disability-adjusted life years (DALYs) from schizophrenia and BD are higher among World Bank high-income countries than other countries. In 2019, the rate of DALYs in high-income countries was 244.18 per 100,000 for schizophrenia and 174.69 for BD. This compares with 195.25 and 100.84 globally [16].

There have been modest, but steady reductions in CVD incidence in the general populations of high-income countries in the last 3 decades [17]. According to the Global Burden of Disease study, incidence of coronary heart disease (CHD) and stroke decreased by 13% and 7%, respectively, in World Bank high-income countries [18] between 1990 and 2019. This may be due to overall improvements in levels of some risk factors, in particular smoking and blood pressure levels [19]. Improved treatments for CVD have also reduced mortality [20].

However, it remains unclear whether the risk of CVD diagnosis and mortality for SMI patients has changed over time.

According to one meta-analysis, the gap in premature mortality between people with SMI and the general population does not seem to have changed over time [2]. It remains unknown whether the gap has remained unchanged also for CVD-related mortality. A recent large cohort study did not describe substantial changes from 2005 to 2010 regarding CVD-related increased mortality in schizophrenia compared with the general population (SMR 3.16 to 3.02), although there was a 10% reduction in SMR for BD (2.08 to 1.87) [3]. However, another cohort study found a widening gap in CVD-related mortality between SMI and the general population [21].

Yet, to the best of our knowledge, a comprehensive systematic review and meta-analysis on the trend over time of increased CVD-related mortality and incidence in schizophrenia and BD is currently missing. Indeed, previous reviews of the association between SMI and CVD have mainly focused on either a single SMI diagnosis (schizophrenia and BD) or a single cardiovascular outcome (myocardial infarction and stroke) or considered CVD mortality separately from incidence, and some excluded unpublished or non-English language studies [5,22–29]. These previous reviews have shown inconsistencies in included studies and differences in the scale and direction of associations. There are also recent large studies not included in previous reviews [30].

Most importantly, none of the previous reviews specifically investigated time trend of CVD-related mortality risk in those with schizophrenia and BD.

An up-to-date, comprehensive, systematic review of the association between SMI and CVD incidence and mortality is therefore indicated. The aim of this systematic review and metaanalysis was to specifically investigate changes over time of CVD-related mortality and incidence, in addition to providing updated pooled estimates of increased CVD mortality and incidence in those with schizophrenia and BD compared with the general population in highincome countries.

Methods

The present systematic review follows a prepublished publicly available protocol registered with PROSPERO (CRD42017058068) and adheres to Preferred Reporting Items for Systematic

Review and Meta-Analyses (PRISMA) guidelines [31] and to Meta-analysis Of Observational Studies in Epidemiology (MOOSE) [32]. PRISMA 2020 checklist and abstract checklist are available in <u>S1</u> and <u>S2 Files</u>, respectively. Minor deviations from the protocol are listed in <u>S3 File</u>.

Inclusion criteria

Population. Eligible studies included participants aged 16 to 65 years at onset of psychotic disorder. The upper age limit was chosen because the onset of psychotic disorders is uncommon at older ages, but transient psychotic conditions are associated with dementia, making it difficult to distinguish between this and SMI [33,34]. Studies were included if at least 90% of participants were within the age range. Only studies in high-income countries (defined as International Monetary Fund "Advanced Economies" [35]: Western Europe, Canada, United States of America, Australia, New Zealand, Taiwan, South Korea, and Japan) were included due to differences between high- and low- or middle-income countries in the diagnosis, prevalence, treatment, and prognosis of SMI [7,8].

Exposure. The exposure of interest was a diagnosis of SMI (schizophrenia, other psychoses, and BD). SMI was defined by the Diagnostic and Statistical Manual of Mental Disorders (DSM), International Classification of Diseases (ICD), and other clinical coding systems (e.g., OPCRIT, OXMIS, and Read codes), operational systems, or clinical diagnosis recorded in patient notes [36–39]. Dissociative, multiple personality disorders, depression (in the absence of psychosis or BD), dementia, learning disabilities, and obsessive-compulsive disorder were excluded.

Comparator. Eligible studies compared people with SMI to control participants without SMI or to the general population. Participants could be in an inpatient, outpatient, or community setting.

Outcome. The primary outcomes were mortality from, or incidence of, CVD (i.e., CHD, cerebrovascular accident (CVA), or heart failure) subsequent to SMI diagnosis and its time trend. Studies that reported composite outcomes including both fatal and nonfatal CVD events were analysed together with incidence studies as the majority of CVD events in these studies were not fatal. Studies were eligible if they described methods to exclude CVD at baseline. However, some studies reporting CVD-related mortality did not exclude individuals with prevalent CVD. These studies were included only if a prevalence of CVD below 10% at baseline was reported. Studies reporting all circulatory disease mortality were included [40] because two-thirds of deaths from all circulatory disease (ICD-10 Chapter I00-I99) are from CHD or stroke. However, those reporting all incident circulatory disease were excluded as incident circulatory disease diagnoses are mainly CVD risk factors [41]. Studies reporting only CVD risk factors were excluded. Studies were eligible if the mean follow-up period was at least 1 year.

Study design. Designs could include cohort studies or case–control studies but not crosssectional, case series, case reports, or uncontrolled studies. Case–control studies compared the odds of SMI compared with no SMI in people who died from, or were newly diagnosed with, CVD compared with matched control participants. See S4 File for full inclusion criteria.

Search strategy

The following bibliographic databases were searched up to November 30, 2021: MEDLINE, EMBASE, and PsycINFO via Ovid and CINAHL via EBSCO. The British Library's ZETOC and Web of Science Conference Proceedings Citation Index were searched to the same date for conference abstracts and the Cochrane Library CENTRAL database searched up to

December 1, 2021 for relevant trials. Initial searches were performed in 2017 and updated in 2019 and again in 2021 using the same search methods but narrowing the searches to 2017 and 2019 onwards, respectively. No language restrictions were applied. Citations of relevant reviews and included studies were searched as was grey literature, including OpenGrey. Detailed search strategies are included in S5 File.

Screening and data extraction

Pairs of reviewers, working independently, firstly screened titles and abstracts and secondly screened full texts. Disagreements were resolved by discussion and if necessary, by reference to a third independent reviewer. Data extraction was conducted by a first reviewer (AML) and checked by a second (EP) using a form adapted from a previous review [5] (S6 File). If necessary, study authors were contacted for more information.

Hazard ratios (HRs), relative risks, rate ratios, risk ratios (RR), odds ratios (ORs), SMRs, or standardised incidence ratios (SIRs) measuring the association between SMI and CVD were extracted or sufficient information to calculate them. Risk measures adjusted for factors related to cardiovascular risk (e.g., socioeconomic factors, comorbidities, smoking, and alcohol use) were extracted where available; otherwise, measures adjusted for age and sex only were extracted. Risk measures were extracted from graphs if not reported in the text. CIs for effect estimates were extracted or calculated from available information.

For studies reporting results for the same cohort that were stratified over calendar time periods, estimates for the whole period were selected if available; otherwise, the earliest time period was selected to avoid a "healthy survivor" effect [30,42–44]. However, for studies that reported SMRs with multiple periods of follow-up [45], actual and expected deaths were summed and the ratio calculated to provide an estimated overall SMR. SMRs for males and females were combined to give a figure for persons if that was unreported. CIs were calculated using the method described by Kirkwood and Sterne [46].

Risk of bias assessment

Risk of bias was assessed using a version of the Newcastle–Ottawa scale (NOS) [47] adapted to include extra criteria on accuracy of outcome measurement and identification of confounders from another checklist for cohort studies [48] (S7 File). Risk of bias was assessed with the NOS independently by 2 researchers. Any discrepancies were discussed, and consensus reached with the aid of a third reviewer when required.

Data synthesis

Narrative and quantitative synthesis of evidence were conducted, and study characteristics recorded in tables for all included studies. Where appropriate, meta-analysis was conducted using DerSimonian and Laird random effects models due to expected population, clinical, and methodological heterogeneity between studies [49]. Heterogeneity was measured using I² and chi-squared statistics and 95% prediction intervals [50]. Associations between different SMIs, CVD outcomes, and type of effect estimate were presented in separate subgroups in forest plots and meta-analyses. Rate ratios were combined with HRs since they have the same interpretation.

Studies that may have included the same patients in overlapping periods of recruitment were not included in the same meta-analyses to avoid double counting outcomes. Rules were devised to determine which studies should be included if the period of overlap was greater than 10% of the total follow-up time (S8 File). Those with fewer exposed cases, or shorter follow-up durations, were excluded to minimise the risk of double counting events.

Where calculable, results for both sexes were included in the meta-analysis; otherwise, separate results for males and females were used. For each meta-analysis with more than 10 studies, potential publication bias or small study effects were assessed visually using funnel plots and formally tested using Egger test [51]. If evidence of significant bias was found, a random effects trim-and-fill procedure [52] was applied to estimate the effect of potential unpublished studies.

To assess temporal trends, studies were allocated to decades according to the median year of follow-up. Further analysis increased the granularity by allocating to 5-year periods according to the median year of follow-up.

All analyses were performed in STATA MP version 14.2 (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, TX: StataCorp LP).

Subgroup analyses

Analysis of prespecified subgroups was conducted. If there were at least 10 study results, random effects meta-regression was performed to explore potential sources of heterogeneity that were measured at study group level [53,54]. Temporal trends, (decade of study conduct), setting (community; community and inpatient; and inpatient), and world region (North America [USA and Canada]); United Kingdom; Scandinavia [Denmark, Finland and Sweden]; Other Western Europe and Israel; Oceania [Australia and New Zealand]; and Far East) were included as covariates in univariate meta-regressions for each subgroup combination of SMI diagnosis (schizophrenia and related conditions; BD), and each mortality outcome (all circulatory disease; CHD; and CVA). Heterogeneity was assessed for each meta-regression by comparing the I² from the random effects meta-analysis and the residual I² after fitting covariates. The amount of between-study variation accounted for by covariates was measured by calculating adjusted R². *p*-Values for model fit were used to assess the association between covariates and SMI and outcomes [55,56]. Subgroup analyses of cohort studies were conducted by sex and by age group at baseline and assessed by forest plots. Studies that reported results for ages ≤ 60 years were plotted separately from results for older groups or all ages.

Sensitivity analyses

Sensitivity analysis was performed to assess the effect of making different decisions on inclusion of studies in the meta-analyses, where studies were subject to overlapping periods of recruitment and thus may have included the same patients, as described above. The effect of inclusion of results for different age groups, medications, or follow-up periods was also assessed in sensitivity analyses.

Sensitivity analysis was conducted to test the effect of replacing included studies with studies that had been excluded due to overlapping recruitment periods. As there was only one case–control study, this was combined with cohort studies.

Results

Main searches identified 16,223 records (both CV incidence and mortality). After removal of duplicates, 11,748 titles and abstracts were screened, and 11,371 were excluded. Of 377 full-text articles assessed, 108 were included in the narrative review. A further 25 studies with over-lapping periods of recruitment and 6 with insufficient information were excluded from the quantitative analysis, leaving 77 included in quantitative analysis of either CVD or CVD mortality. Of these, 59 reported CV mortality data, and 28 reported CV incidence (Fig 1). Mortality studies included in the narrative review are listed in <u>S9 File</u> and incidence studies in <u>S10 File</u>. Studies excluded from the narrative review are listed, with reasons, in <u>S11 File</u>.



Fig 1. PRISMA flowchart. CVD, cardiovascular disease; SMI, severe mental illness.

https://doi.org/10.1371/journal.pmed.1003960.g001

Characteristics of included studies

Mortality. All 82 studies included in the review reporting mortality outcomes were cohort studies (S12 File). Of these, 66 included patients with schizophrenia or related conditions, 30 included patients with BD, and 6 articles reported results for mixed SMI. For CVD mortality outcomes, 43 reported CHD, 31 reported CVA, 1 reported heart failure, and 54 reported all circulatory disease mortality. SMRs were the most common measure of relative risk (48 papers). The others reported rate ratios or HRs apart from a single study that reported ORs. Combined results for both sexes were obtained from most articles (k = 67). A total of 15 articles contributed outcomes from the 1970s or earlier, 18 from the 1980s, 20 from the 1990s, 29 from the 2000s, and 7 since 2010. Papers may have contributed results to more than 1 decade. Moreover, 38 of the 82 mortality studies were from Scandinavia, 13 from North America, 12 from Western Europe and Israel (excluding UK and Scandinavia), 11 from the UK, 5 from the Far East, and 3 from Oceania. A total of 41 studies were inpatient only; 17 were community only; and 24 were conducted in both inpatient and community settings. Furthermore, 5 studies that did not include CIs or standard errors were not included in the meta-analysis.

Incidence. CVD incidence was reported by 40 studies (S13 File): 1 case–control study and 39 cohort studies. Altogether, 29 articles included patients diagnosed with schizophrenia or related conditions and 21 included patients with BD. No studies reported results for mixed SMI. For cardiovascular outcomes, CHD was reported by 28 articles, CVA by 18, heart failure by 5, and incidence of total CVD events, defined mainly as combined heart disease and stroke, by 9. Rate ratios were reported by 13, HRs by 14, and ORs by 6; relative risks were obtained from 6 studies, and 1 reported SIRs. Results for all persons were reported by 33 articles, 6 reported males and females separately and 1 reported results for males only. No studies reported outcomes with a median decade of the 1970s or earlier; 6 reported results from the 1980s, 11 from the 1990s, 23 from the 2000s, and 1 since 2010. Moreover, 18 of the incidence studies were from Scandinavia; 10 from North America; and 7 were from the Far East. The others were from the UK, Australia, other Western Europe (excluding Scandinavia and UK), and Israel. A total of 12 studies were inpatients only, 19 were both inpatients and community, and 9 were set in the community only. One study [57] did not report CIs or standard errors, so we were unable to include those results in the meta-analysis.

Full forest plots for all studies included in the review are in <u>S14</u> and <u>S15</u> Files.

Risk of bias of included studies

Mortality. Most studies that reported mortality outcomes were rated to have high risk of bias mainly due to lack of adjustment for confounders, lack of clarity about presence of CVD at baseline, and control participants being drawn from a different population to the SMI cohort (S16 File). Many studies assessed as being of higher risk of bias used general population mortality rates as the comparison. Only half the studies included persons with SMI in the community; the other half included only inpatients. Ascertainment of SMI was good as most studies used medical records. However, there was potential for bias due to a lack of clarity regarding ascertainment of all cardiovascular outcomes. Adequacy of reporting and duration of follow-up were rated as good for around half of the studies.

Incidence. There was also high risk of bias among many of the included studies that reported CVD incidence (S17 File); ascertainment of CVD was good in only 7 cohort studies, and there was a lack of clarity regarding presence of CVD at baseline for 32 studies. However, 26 incidence studies adjusted for at least 1 confounder, and adequacy of reporting and duration of follow-up were rated as good for over half of the studies. Overall, 11 incidence studies [58–68] were assessed to have low risk of bias.

Meta-analysis

Mortality. Altogether, 59 studies contributed HRs, rate ratios, ORs, or SMRs of cardiovascular mortality in people with SMI for meta-analysis. Some studies contributed more than 1 result, e.g., different SMIs, multiple outcomes, or separate male and female results. The number of included participants was not reported for some studies; however, the total among those that did report this information was 1,841,356 cases and 29,321,409 controls. The risk of CVD mortality for people with SMI was higher than for control participants, for each type of SMI, each CVD outcome, and each type of effect, except pooled HRs/rate ratios for all circulatory disease mortality in BD and SMRs for CVA in mixed SMI. Pooled estimated risks appeared larger for schizophrenia than BD. Pooled results for SMI/CVD combinations with more than 1 study ranged from SMR = 1.55 (95% CI: 1.33 to 1.81, p < 0.001) for CVA in people with BD to HR/rate ratio = 2.40 (95% CI: 2.25 to 2.55, p < 0.001) for CVA in schizophrenia (Table 1).

We found significant heterogeneity affecting all subgroups (p < 0.05 and $I^2 > 50\%$), except for CVA pooled HRs/rate ratios in both schizophrenia and BD and ORs for CVD in BD (the latter based on a single study). Prediction intervals may be informative in describing the likely range of future studies [54]. For schizophrenia, 95% prediction intervals did not overlap unity for CVA, and for BD, they did not overlap unity for CHD and all circulatory disease SMRs. Thus, for these combinations of SMI and CVD mortality, we would expect a future study to report elevated mortality ratios. Figs 2 and 3 show forest plots of studies reporting CVD mortality outcomes for schizophrenia and BD that were included in meta-analyses.

SMI	Outcome	Measure	Studies/Results	Pooled effect (95% CI)	<i>p</i> -value	Prediction interval	I ²	X ²	df	<i>p</i> -value
SCZ	CVA	HR/rate ratio	3/3	2.4 (2.25–2.55)	< 0.001	1.6-3.6	0.0	0.5	2	0.785
SCZ	CVA	SMR	14/16	1.93 (1.63–2.28)	< 0.001	1.02-3.67	91.5	176.3	15	< 0.001
SCZ	CHD	HR/rate ratio	7/10	1.8 (1.44–2.24)	< 0.001	0.79-4.09	98.9	827.1	9	< 0.001
SCZ	CHD	SMR	14/16	1.94 (1.58–2.37)	< 0.001	0.81-4.62	98.0	733.8	15	< 0.001
SCZ	CVD	HR/rate ratio	10/12	1.91 (1.52–2.41)	< 0.001	0.79-4.66	98.9	979.0	11	< 0.001
SCZ	CVD	OR	1/2	2.55 (1.74-3.74)	< 0.001		91.8	12.1	1	< 0.001
SCZ	CVD	SMR	18/20	1.96 (1.61–2.39)	< 0.001	0.77-4.97	99.2	2414.0	19	< 0.001
SCZ	HF	HR/rate ratio	1/1	3.25 (2.94-3.6)	< 0.001			0.0	0	
BD	CVA	HR/rate ratio	2/2	2.01 (1.81–2.22)	< 0.001		0.0	0.5	1	0.498
BD	CVA	SMR	5/7	1.55 (1.33-1.81)	< 0.001	0.99-2.44	74.0	23.1	6	0.001
BD	CHD	HR/rate ratio	5/7	1.61 (1.34–1.94)	< 0.001	0.93-2.8	90.2	61.2	6	< 0.001
BD	CHD	SMR	5/7	1.67 (1.54–1.82)	< 0.001	1.33-2.1	57.7	14.2	6	0.028
BD	CVD	HR/rate ratio	2/2	1.52 (0.84–2.75)	0.162		94.6	18.5	1	< 0.001
BD	CVD	OR	1/2	1.83 (1.57–2.14)	< 0.001		65.2	2.9	1	0.090
BD	CVD	SMR	12/14	1.65 (1.53–1.77)	< 0.001	1.29-2.1	80.4	66.4	13	< 0.001
SMI	CVA	SMR	1/1	0.8 (0.16-4.08)	0.788			0.0	0	
SMI	CHD	SMR	1/1	3.7 (2.47–5.55)	< 0.001			0.0	0	
SMI	CVD	HR/rate ratio	1/1	5.41 (2.4–12.2)	< 0.001			0.0	0	
SMI	CVD	SMR	4/5	2.3 (1.63–3.25)	< 0.001	0.63-8.4	92.9	56.3	4	< 0.001

Table 1. Pooled relative risks of cardiovascular mortality by SMI and cardiovascular outcome.

"."—not calculable (insufficient observations).

Results where 95% CIs exclude the null highlighted in bold.

BD, bipolar disorder; CHD, coronary heart disease; CI, confidence interval; CVA, cerebrovascular accident; CVD, all circulatory disease; HF, heart failure; HR, hazard ratio; OR, odds ratio; SCZ, schizophrenia; SMI, mixed schizophrenia and bipolar disorder; SMR, standardised mortality ratio.

https://doi.org/10.1371/journal.pmed.1003960.t001



Fig 2. Forest plots showing relative risk of CVD mortality in those with versus without schizophrenia, studies included in meta-analysis. Weights are from random effects model. Dashed lines show pooled estimated risks within subgroups. Studies were allocated to decades according to the median year of follow-up. CHD, coronary heart disease; CVA, cerebrovascular accident; CVD, all circulatory disease; F, female; HR, hazard ratio; M, male; OR, odds ratio; SCZ, schizophrenia; SMR, standardised mortality ratio.

https://doi.org/10.1371/journal.pmed.1003960.g002

Incidence. Results from 28 studies that contributed data on risk of cardiovascular incidence in people with SMI compared with control participants were used in meta-analysis. Some studies contributed more than 1 result. The number of included participants for studies that did report this information was 401,090 cases and 14,372,146 controls. Pooled results for schizophrenia and BD, for incidence of CVA, CHD, major CVD events, and heart failure, are shown in Table 2. As



Fig 3. Forest plots showing relative risk of CVD mortality in those with versus without BD, studies included in meta-analysis. Weights are from random effects model. Dashed lines show pooled estimated risks within subgroups. Studies were allocated to decades according to the median year of follow-up. BD, bipolar disorder; CHD, coronary heart disease; CVA, cerebrovascular accident; CVD, all circulatory disease; F, female; HR, hazard ratio; M, male; OR, odds ratio; SMR, standardised mortality ratio.

https://doi.org/10.1371/journal.pmed.1003960.g003

SMI	Outcome	Measure	Studies/Results	Pooled effect (95% CI)	<i>p</i> -value	Prediction interval	I ²	X ²	df	<i>p</i> -value
SCZ	CVA	HR/rate ratio	4/5	1.66 (1.36–2.02)	< 0.001	0.82-3.34	94.6	74.2	4	< 0.001
SCZ	CVA	OR	2/2	1.88 (1.31–2.7)	0.001		0.0	0.3	1	0.586
SCZ	CVA	RR	3/3	1.4 (1.31–1.49)	< 0.001	0.94-2.09	0.0	0.3	2	0.861
SCZ	CHD	HR/rate ratio	8/10	1.15 (0.99–1.35)	0.077	0.65-2.04	97.8	415	9	< 0.001
SCZ	CHD	OR	2/2	0.76 (0.61–0.96)	0.022		0.0	0.4	1	0.515
SCZ	CHD	RR	3/3	0.91 (0.72-1.14)	0.403	0.07-11.47	73.7	7.6	2	0.022
SCZ	CHD	SIR	1/2	1.48 (1.3–1.69)	< 0.001		85.0	6.7	1	0.01
SCZ	CVD	HR/rate ratio	3/3	1.25 (1.04–1.51)	0.016	0.14-11.5	83.3	12	2	0.003
SCZ	CVD	OR	1/2	1.8 (1.24–2.62)	0.002		95.3	21.4	1	< 0.001
SCZ	HF	HR/rate ratio	1/1	3.82 (3.1–4.71)	< 0.001			0.0	0	
SCZ	HF	OR	2/2	1.79 (1.04–3.07)	0.034		57.3	2.3	1	0.126
SCZ	HF	RR	2/2	1.72 (1.08–2.75)	0.023		75.1	4.0	1	0.045
BD	CVA	HR/rate ratio	4/5	1.6 (1.41–1.81)	< 0.001	1.07-2.4	70.6	13.6	4	0.009
BD	CVA	OR	1/1	3.39 (1.91–6.01)	< 0.001		0.0	0.0	0	
BD	CVA	RR	2/2	1.31 (1.18–1.45)	< 0.001		0.0	0.0	1	0.882
BD	CHD	HR/rate ratio	5/6	1.47 (1.16–1.87)	0.002	0.63-3.43	94.6	92.2	5	< 0.001
BD	CHD	OR	1/1	2.68 (1.51–4.76)	0.001			0.0	0	
BD	CHD	RR	1/1	0.97 (0.5–1.88)	0.928			0.0	0	
BD	CVD	HR/rate ratio	2/2	1.47 (1.32–1.63)	< 0.001		0.0	0.1	1	0.807
BD	CVD	OR	2/3	1.88 (1.34–2.66)	< 0.001	0.03-134.7	95.4	43.7	2	< 0.001
BD	HF	HR/rate ratio	1/1	3.3 (2.27–4.79)	< 0.001			0.0	0	
BD	HF	RR	1/1	1.73 (0.88-3.4)	0.111			0.0	0	

Table 2. Pooled relative risks of cardiovascular incidence by SMI and cardiovascular outcome.

"."—not calculable (insufficient observations).

Results where 95% CIs exclude the null highlighted in bold.

BD, bipolar disorder; CHD, coronary heart disease; CI, confidence interval; CVA, cerebrovascular accident; CVD, major cardiovascular events; HF, heart failure; HR, hazard ratio; OR, odds ratio; RR, risk ratio; SCZ, schizophrenia; SIR, standardised incidence ratio.

https://doi.org/10.1371/journal.pmed.1003960.t002

for mortality, most pooled results involving more than 1 study indicated an elevated risk of CVD incidence for participants with SMI compared with control participants, with the notable exception of CHD for people with schizophrenia. Pooled ORs indicated a reduced risk of CHD incidence (OR: 0.76, 95% CI: 0.61 to 0.96, p = 0.077), while RRs and pooled HRs/rate ratios included the null. Risks of CVA, total cardiovascular events, and heart failure for both schizophrenia and BD and risk of CHD for BD were in a positive direction for all except 2 studies, which were not significantly different from unity [69,70]. In subgroups containing more than 1 study, there was significant heterogeneity in most groups (p < 0.1) and $I^2 > 50\%$, except for ORs and RRs for CVA and ORs for CHD and HF in schizophrenia, CVA RRs, and CVD HR/rate ratios in BD. Prediction intervals for all subgroups included unity except for HRs/rate ratios for CVA in BD. Studies reporting incidence outcomes are shown in Figs A and B in S18 File.

Temporal trends

Mortality. Results from meta-regression (Table 3) showed that for schizophrenia, SMRs for cerebrovascular mortality in the 1990s were 2-fold higher than in the 1970s and earlier (SMR: 1.96, 95% CI: 1.02 to 3.78, p = 0.044). The earliest decade for which results were available was chosen as the reference group. Pooled HR/rate ratios for CHD were 60% higher in the 1990s compared with the 1980s (HR/rate ratio: 1.61 [95% CI: 1.14 to 2.28, p = 0.014]). However, the estimated risks for the strength of association between SMI and CVD was lower

Explanatory variable (if no. of studies/results \geq 10)				BD		
			CVA	CHD	CVD	CVD
Decade	SMR	1950s-70s (ref) 1980s 1990s 2000s 2010s	1 1.14 (0.59–2.20), 0.673 1.96 (1.02–3.78), 0.044 1.45 (0.85–2.46), 0.153 1.01 (0.54–1.89), 0.978	1 0.67 (0.36–1.22), 0.169 1.38 (0.75–2.52), 0.267 1.55 (0.92–2.59), 0.089 1.31 (0.70–2.45), 0.358	1 0.43 (0.17–1.06), 0.063 1.26 (0.63–2.53), 0.489 1.51 (0.77–2.96), 0.214 1.52 (0.70–3.29), 0.271	1
	Model p-value, residual I^2 , adjusted R^2		0.193, 79.0%, 37.5%	0.073, 96.6%, 38.3%	0.059, 98.3%, 39.5%	0.003, 34.3%, 88.9%
	HR/rate ratio	1950s-70s 1980s (ref) 1990s 2000s 2010s	Insufficient observations	1 1.61 (1.14–2.28), 0.014 0.75 (0.56–1.00), 0.052	- 1 1.99 (0.46–8.60), 0.314 1.54 (0.39–6.14), 0.500	Insufficient observations
	Model p-value, residual I^2 , adjusted R^2			0.007, 87.0%, 85.7%	0.487, 98.3%, 2.1%	
Setting	SMR	Community (ref) Community and inpatient Inpatient	1 0.62 (0.21–1.81), 0.355 0.61 (0.21–1.74), 0.325	1 1.02 (0.37–2.80), 0.969 0.76 (0.28–2.03), 0.552	1 1.19 (0.43–3.29), 0.728 0.93 (0.33–2.58), 0.887	1 0.96 (0.77–1.19), 0.672 -
	Model p-value,	residual I ² , adjusted R ²	0.603, 92.5%, -8.8%	0.412, 97.4%, -0.9%	0.699, 98.6%, -5.7%	0.672, 81.5%, -11.1%
	HR/rate ratio	Community (ref) Community and inpatient Inpatient	Insufficient observations	1 0.67 (0.26–1.76), 0.359 0.98 (0.47–2.05), 0.954	1 1.25 (0.79–1.98), 0.292 1.71 (0.79–3.68), 0.150	Insufficient observations
	Model <i>p</i> -value, residual I^2 , adjusted R^2			0.524, 99.1%, -18.7%	0.279, 98.2%, 6.1%	
Region	SMR	UK (ref) Far East North America Oceania Other EU/Israel Scandinavia	1 0.32 (0.15–0.68), 0.007 0.58 (0.27–1.24), 0.140 0.95 (0.42–2.13), 0.888 0.58 (0.28–1.19), 0.122 0.60 (0.30–1.17), 0.118	1 0.78 (0.25–2.38), 0.626 1.11 (0.36–3.39), 0.836 1.09 (0.29–4.11), 0.887 0.78 (0.29–2.13), 0.600 0.95 (0.37–2.43), 0.897	1 0.88 (0.31–2.46), 0.787 1.26 (0.46–3.42), 0.634 1.98 (0.49–8.07), 0.313 0.63 (0.23–1.75), 0.351 1.35 (0.58–3.15), 0.457	1 1.13 (0.79–1.64), 0.451 1.06 (0.72–1.56), 0.740 1.31 (0.79–2.16), 0.255 1.11 (0.70–1.75), 0.620 1.26 (0.92–1.71), 0.131
	Model p-value, residual I^2 , adjusted R^2		0.034, 68.5%, 59.9%	0.942, 94.4%, -37.6%	0.417, 97.9%, 0.04%	0.586, 73.2%, -17.4%
	HR/rate ratio	UK (ref) Far East North America Oceania Other EU/Israel Scandinavia	Insufficient observations	1 0.38 (0.11–1.30), 0.103 0.36 (0.11–1.21), 0.085 0.52 (0.18–1.51), 0.187	1 - 1.11 (0.56–2.21), 0.729 - 1.51 (0.67–3.42), 0.273 1.80 (0.88–3.67), 0.094	Insufficient observations
	Model p-value, residual I^2 , adjusted R^2			0.247, 99.2%, 16.6%	0.109, 94.8%, 53.8%	
Random effects meta- analysis I ²		SMR HR/rate ratio	91.5%	98.0% 98.9%	99.2% 98.9%	80.4%

Table 3. Results of meta-regressions: estimates of increased risk relative to the reference category for study-level explanatory variables; exponentiated regression coefficients (95% CIs), *p*-values.

BD, bipolar disorder; CI, confidence interval; CHD, coronary heart disease; CVA, cerebrovascular accident; CVD, all circulatory disease; HR, hazard ratio; North America, USA and Canada; Other EU, Western Europe excluding UK and Scandinavia; ref, reference group; SMR, standardised mortality ratio. Results where 95% CIs exclude the null highlighted in bold. Models adjusted for age and sex and minimum number of additional factors.

https://doi.org/10.1371/journal.pmed.1003960.t003

in the 2000s compared with the 1980s (HR/rate ratio: 0.75 [95% CI: 0.56 to 1.00, p = 0.052]). BD studies in the 2000s reported SMRs for all circulatory disease higher than those in the 1950s to 1970s (SMR: 1.27, 95% CI: 1.09 to 1.47, p = 0.006).

Heterogeneity between studies was reduced by the inclusion of decade in the case of schizophrenia and CVA ($I^2 = 91.5\%$, residual $I^2 = 79.0\%$), with decade accounting for a substantial amount of the between study variation (adjusted $R^2 = 37.5\%$). Similarly, for BD and all CVD mortality, decade reduced heterogeneity and accounted for considerable between-study variation ($I^2 = 80.4\%$, residual $I^2 = 34.3\%$, adjusted $R^2 = 88.9\%$). For schizophrenia and CHD, there was still considerable heterogeneity after the inclusion of decade (residual $I^2 = 96.6\%$, adjusted



Fig 4. Trend in risk of CVD mortality for SMI compared with control participants, by median decade. BD, bipolar disorder; CHD, coronary heart disease; CI, confidence interval; CVA, cerebrovascular accident; CVD, all circulatory disease; HR, hazard ratio; RRatio, rate ratio; SMI, severe mental illness; SMR, standardised mortality ratio; SCZ, schizophrenia. The year 1970 includes studies with median between 1950 and 1970. y-Axes show estimated risks on log scale, and shaded areas show 95% CIs for risk estimates.

https://doi.org/10.1371/journal.pmed.1003960.g004

 $R^2 = 85.7\%$), but the model *p*-value of 0.007 indicates some evidence for a relationship between decade and size of association between schizophrenia and CVD mortality.

Fig 4 shows the trends in CVD mortality by SMI across decades. For schizophrenia and BD, there appeared to be stronger associations between SMI and both CVD and CHD mortality in studies reporting SMRs occurring during the 1990s and 2000s, compared with earlier decades, suggesting that, on this measure, the gap between people with SMI and the general population may be widening. In the 2010s, SMRs were reduced slightly, but CIs were wider. Pooled HRs and rate ratios appeared to peak in the 1990s. CVA SMRs peaked in the 1990s for schizophrenia and in the 2000s for BD. A summary forest plot of meta-analysis results by SMI, outcome, and decade with the inclusion of prediction intervals is shown in Fig A in S19 File. Most prediction intervals were wide and included the null, indicating that any future study could potentially show a wide range of potential effects including decreased risk of CVD mortality for SMI compared with control participants (see also Table A in S19 File).

For most SMI/CVD combinations, increasing the granularity by splitting time into 5-year periods for meta-regression did not add any further information as there were too few studies





https://doi.org/10.1371/journal.pmed.1003960.g005

in each group. For those groups that did have sufficient observations, CIs were wide, and most results indicated no increased risk for any 5-year period compared with the reference period. Only for CVA in schizophrenia did the 5-year analysis show higher SMRs for 1995 to 1999 compared with 1950 to 1969 (see Table C in <u>S19 File</u>). Graphs showing the trends in CVD-related mortality for SMI compared with control participants, by 5-year periods, are presented in Fig C in <u>S19 File</u>.

Incidence. Meta-regression was not performed on cardiovascular incidence data as there were insufficient results in each subgroup. Fig 5 shows the trends in CVD incidence by SMI across decades. For schizophrenia and BD, RRs and pooled HRs and rate ratios increased during the 2000s. No studies reported results on these measures beyond this decade. These measures appear to show a widening gap between people with SMI and the general population. ORs, however, showed decreased effects between the 1990s and later years.

A summary forest plot of meta-analysis results by SMI, CVD incidence outcome, and decade with prediction intervals is shown in Fig B in <u>S19 File</u>. All prediction intervals that were able to be calculated included the null, indicating that any future study could potentially

show a decreased risk of CVD incidence for SMI compared with the general population. Tabulated results and trends by 5-year intervals are in Table B and Fig D in <u>S19 File</u>.

Subgroup analysis

Mortality. Age group. Estimated risks for younger age groups were larger than for older ages in the same study or in the equivalent SMI/outcome/effect type subgroup (Fig A in S20 File).

Sex. CIs of pooled estimates for males and females overlapped for all combinations of SMI and CVD mortality (Figs B, C and D in <u>S20 File</u>), except for schizophrenia ORs for CVD mortality and pooled HRs/rate ratios for heart failure mortality. Single studies contributed results to these 2 outcomes. There was a higher reported estimated risk for mortality from heart failure for males compared with females (males: rate ratio: 4.30, 95% CI: 3.71 to 4.98; females: rate ratio: 2.66, 95% CI: 2.32 to 3.06). However, the OR for circulatory mortality was higher for females than males (males: OR 2.13, 95% CI: 2.04 to 2.23; females: OR 3.15, 95% CI: 2.54 to 3.91).

Location and setting. Results of meta-regression showed no differences in CVD-related mortality for any region compared with the UK, except for studies reporting SMRs for the Far East, which had a smaller association between schizophrenia and CVD mortality: SMR: 0.32, 95% CI: 0.15 to 0.68, p = 0.007 (Table 3). However, there was still considerable residual heterogeneity between studies (residual I² > 68.5%). The inclusion of region reduced I² from 91.5% to 68.5%.

The association between SMI and CVD did not vary for different settings (inpatient, outpatient, and community).

Other factors. No further subgroup analyses by study quality, length of follow-up, or confounding factors were completed owing to lack of applicable studies. Only 3 mortality studies were judged to be of low risk of bias and studies either lacked reported rates for confounding factors in the SMI and control populations or varied in the confounders included.

Incidence. Risk of bias. Meta-analysis was conducted for the studies assessed to be at low risk of bias. Pooled relative risks for the low risk of bias studies presented a similar picture to results from studies included in the meta-analysis for most SMI/CVD subgroups (see Fig E in S20 File). One notable exception was incidence of CHD for schizophrenia. The pooled HR/ rate ratio for the higher-quality schizophrenia studies was raised (HR/rate ratio: 1.35, 95% CI: 1.03 to 1.79, p = 0.032), whereas the pooled result for all included schizophrenia studies did not exclude the null (HR/rate ratio: 1.15, 95% CI: 0.9 to 1.35, p = 0.077).

Sensitivity analysis

Excluded studies. A total of 18 mortality studies and 11 incidence studies were excluded from the meta-analysis because overlapping periods of exposure with another included study may have meant that the same participants were double counted if the studies were included. Sensitivity analyses explored the effect of replacing included studies with excluded ones. This made little difference to the findings for CVD mortality; however, some pooled estimated risks were increased (Table A in S21 File). Only one inclusion affected results, increasing the rate ratio of circulatory mortality for BD from HR/rate ratio: 1.52 (95% CI: 0.84 to 2.75) to 1.78 (95% CI: 1.28 to 2.47) when the included study [71] was replaced by an alternative that used the same exposed population [60]. Sensitivity analysis made little difference to the findings for BD and CVD incidence (Table B in S21 File). However, for schizophrenia, there were inconsistent results when substituting some of the included studies, particularly for CHD. The pooled

HR/rate ratio for CHD was borderline for the studies included in the meta-analysis (HR/rate ratio: 1.15 (95% CI: 0.99 to 1.35). Sensitivity analysis widened the CIs around the null for some studies but increased the estimated risk away from the null for others.

There were 2 excluded studies of schizophrenia that we were not able to obtain full texts for, so could not confirm the methodology, but they showed similar results for incidence of CHD [72] and heart failure [73] to those presented in Table 1. Another study excluded from the review, because middle-income as well as high-income countries were included [74] reported similar results for incidence of stroke to those in Table 2.

Confounding factors (incidence). Results adjusted for age and sex were extracted from incidence studies. A total of eight studies reported results adjusted by age and sex only. Moreover, 20 studies reported models adjusted for up to 18 additional factors, median 5 factors. Of these, 5 did not report results adjusted for age and sex alone. Meta-analysis using models adjusted for factors in addition to age and sex made little difference to the direction of results, although estimated risks were reduced for some subgroups (Table C in S21 File).

Seven incidence studies adjusted for smoking. Seven studies reported rates of antipsychotic drug use. Subgroup analysis by confounding factors was not possible, because there was too much variability in the confounders adjusted for.

Publication bias and small study effects. Small study effects were assessed visually by funnel plots (S22 and S23 Files). Mortality results for SMRs, ORs, HRs, and rate ratios were combined for visual assessment. There was some asymmetry in the funnel plots, particularly for CHD and CVD mortality in schizophrenia. Contour funnel plots for these 2 subgroups showed that some studies with nonsignificant results may be missing (white areas in the plots). Egger test was performed for subgroups with more than 10 studies (Table A in S22 File). There was evidence of significant biases for CHD mortality and CVD mortality in schizophrenia studies reporting SMRs. Applying random effects trim-and-fill procedures [53] added 5 unpublished studies reporting CHD mortality and 2 reporting all circulatory disease mortality for schizophrenia (Figs G and H in S22 File). The resulting pooled random effects estimate changed from SMR = 1.94 (95% CI: 1.58 to 2.37, p < 0.001) to SMR = 1.61 (95% CI: 1.11 to 2.33, p = 0.012) for CHD and from SMR = 1.96 (95% CI: 1.61 to 2.38, p < 0.001) to SMR = 1.77 (95% CI: 1.27 to 2.47, p = 0.001) for all circulatory disease.

Incidence results for HRs, rate ratios, ORs, RRs, and SIRs were combined due to small numbers of studies in each group. Egger test was performed for incidence of CVA and CHD in schizophrenia as only those subgroups had more than 10 studies (Table A in S23 File). Significant small study bias was not found when different SMIs and different outcomes were considered separately, although when results from all studies were considered together, publication bias was indicated (Egger test for bias: 1.94, 95% CI: 0.16 to 3.72, p = 0.034). However, applying trim and fill methodology did not add any additional unpublished studies.

Discussion

This large-scale systematic review has been conducted in a methodologically robust way and has yielded up-to-date comprehensive evidence on the association between SMI- and CVD-related mortality and incidence. It has found that the CVD mortality rate for people with SMI is around twice that of the general population. People with schizophrenia are at greater risk than those with BD; however, the disparity exists across all types of SMI and both cerebrovas-cular and cardiac mortality. The review also found a consistent relationship between SMI and increased incidence of CVA, CVD events, and heart failure. The relationship between SMI and CHD was less clear; a positive association only became apparent for high-quality studies with low risk of bias.

Some evidence was found of changes in the risk of CVD over time, and results from some studies suggest that the gap between SMI and the general population appears to be widening. The size of the association between SMI and both CHD and all circulatory mortality increased until the 2000s, following this it appeared to decline. Estimated risks of the associations between schizophrenia and CVA mortality were also highest in the 1990s, whereas for BD, they were highest for CVA mortality in the 2000s.

Studies reporting incidence of CVD since the 1990s had larger associations on most measures between SMI and diagnosis of CVA and CHD compared with earlier studies. ORs appeared to show the reverse trend; however, only 5 studies contributed results that could be used in this trend analysis.

The apparent increases in CVD risk for people with SMI relative to control participants that we have found in recent decades are in marked contrast to the reducing rates of CVD experienced by the general population. Since 1990, there has been a steady decline in the rates of CVD mortality and incidence in the populations of high-income countries [17]. Most European countries have also seen a halving in overall population CHD mortality since the 1980s [75].

Increasing trends in CVD mortality for people with SMI relative to control participants have been noted by other authors [30,44,76,77]. An increase in CHD hospital admissions for people with schizophrenia compared with the general population since the 1990s has also been reported [44].

The increased relative risk of CVD diagnosis in more recent decades may be a result of disparity in smoking prevalence between people with SMI and the general population [78,79] or increased use of antipsychotics. The changes since the 1990s approximately coincide with the release of newer, second-generation antipsychotics (SGAs), which are known to have worse metabolic effects [80–82]. The use of these drugs has increased over the last 20 years for treatment of both schizophrenia and BD [83–86].

There was significant heterogeneity between studies, which was partly explained by decade of results and world region, the latter due to differences in CVA incidence between the Far East and the UK. Although some evidence of differences between decades and between regions were found, the large number of CIs calculated from multiple meta-regressions mean that findings should be interpreted cautiously. Some evidence of small study effects was found by the review, affecting the association between schizophrenia and CVD mortality. However, publication bias is unlikely to account for the increased risk of CVD mortality found in the review as the pooled effect remained elevated, even after application of the trim and fill method. Small study bias could only be assessed for incidence studies when both types of SMI exposure and all CVD outcomes were combined. Although significant bias was found, analysis found no additional missing studies. This analysis may be flawed as it was not possible to exclude double counting of populations at risk or of outcomes. Thus, results may not have been independent, and the random effects assumption may have been violated. However, the analysis is of value in demonstrating that the scale of any potential publication bias is likely to be small.

Comparison with previous literature

Mortality. Schizophrenia

<u>Cerebrovascular accident</u>. In contrast to all circulatory disease and CHD, the pooled CVA mortality ratio for schizophrenia, SMR: 1.93 (95% CI: 1.63 to 2.28, p < 0.001), is in the opposite direction to that found by other authors [24,28]. It is unclear which studies were included in these reviews, but the reviews were relatively small: One review only included 3 studies (median SMR: 0.69, IQR: 0.61 to 1.30) [24], and the other reported an SMR based on only 490 deaths (SMR: 81, 95% CI: 73 to 87) [28].

The current review, however, included 14 studies in the pooled SMR; it was based on 2,800 deaths, and all studies showed a higher risk of CVA mortality for people with schizophrenia compared to the general population. Pooled HRs and rate ratios were also elevated and were not reduced by the substitution of excluded studies in sensitivity analyses.

Coronary heart disease. The pooled result for the 14 studies reporting CHD SMRs for schizophrenia, SMR: 1.94 (95% CI: 1.58–2.37, p < 0.001), was higher, but in the same direction as other smaller reviews [24,28]. Further, sensitivity analysis showed that substituting some of the excluded studies could change the pooled SMR and HR/rate ratio. Small study effects or publication bias could account for some of this discrepancy; an analysis accounting for small study bias reduced the estimated risk to SMR: 1.61 (95% CI: 1.11 to 2.33, p = 0.012).

<u>All circulatory disease</u>. The pooled results for circulatory mortality in schizophrenia, from 28 studies (HR/rate ratio: 1.91, 95% CI: 1.52 to 2.41, p < 0.001; SMR: 1.96, 95% CI: 1.61 to 2.39, p < 0.001), were slightly lower than, but in broad agreement with, those of a recent review based on 9 studies [5]. Sensitivity analysis found that rate ratios were increased by the inclusion of one study that had high HRs [87] that was also included in the previous review [5].

Bipolar disorder

<u>All circulatory disease</u>. The pooled SMR of 1.65, 95% CI: 1.53 to 1.77, p < 0.001 and combined HR and rate ratio of 1.52, 95% CI: 0.84 to 2.75, p = 0.162 for all circulatory mortality in BD are similar to ratios reported by other authors [5,27].

Incidence. Schizophrenia

<u>Cerebrovascular accident</u>. Previous reviews have also reported increased risks of stroke with schizophrenia and with bipolar disease [5,22,23,25].

<u>Coronary heart disease</u>. The current review found that people with schizophrenia are less likely to be diagnosed with CHD than the general population, in contrast to previous reviews [5,25]. The disparity may be explained by variation in included primary studies.

Heterogeneity in incidence of CHD could reflect differences in ascertainment. Five schizophrenia studies only included inpatient diagnosis of CHD [65,88–91], which could have missed both milder cases treated in the community and deaths from undiagnosed CHD outside of hospital.

Bipolar disorder

<u>Coronary heart disease</u>. The nonsignificant positive associations between CHD and BD were similar to those reported elsewhere [5,22].

Schizophrenia and bipolar disorder

<u>All circulatory disease</u>. The increased likelihood of total CVD events and SMI (both schizophrenia and BD) correspond with the findings of previous reviews [5,25].

Change over time. We have found evidence that the gap in CVD mortality between people with SMI and the general population appears to be widening. This finding supports the observations in some of the included studies that CVD mortality has increased in recent years for people with SMI relative to the general population [21,44,76,77]. This is likely to be due to a number of factors. In developed countries, adult smoking rates have fallen for the general population in recent decades: from average rates of 33.1% in 1980 to 23.5% in 2012 [92]. However, people with long-term mental health problems have not seen a similar decline [78,79]. SGA medication has been in use since the 1990s [81,82]. However, there is a possible trade-off in their use between the benefits of treating psychotic symptoms against the metabolic problems that can result which increase the risk of CVD. Use of some SGAs, especially clozapine and olanzapine, is likely to result in weight gain, particularly with long-term use [93]. Certain SGAs (particularly clozapine, olanzapine, and quetiapine) are associated with a higher risk of diabetes than first-generation antipsychotics (FGAs) [94]. There is also some evidence of elevated lipid levels with use of clozapine, olanzapine, and risperidone compared with FGAs [95].

Differential access to physical health treatments for people with SMI is also an issue. Rates of cardiac revascularisation procedures have increased over the last 20 years [96]. However, people with schizophrenia who have acute coronary syndrome are much less likely than others to receive this type of treatment [8].

Strengths and limitations

Strengths. The strengths of this review are its size and scope. Both schizophrenia and BD and different cardiovascular diagnoses were included. A wide search strategy was used, not limited by language or date of publication. Studies reporting different measures of relative risk were included, and changes over time were investigated. Robust methodology including duplicate screening, and data extraction was used. Possible sources of heterogeneity between studies were investigated. Most studies had an average follow-up duration of at least 5 years, some considerably longer, and long enough for CVD outcomes to occur. The adequacy of reporting follow-up was generally good as many of the studies were carried out in Scandinavia, which has near complete recording of population health records.

Limitations. There are also several limitations. It was difficult to exclude studies that had overlapping populations. Although attempts were made to minimise this, it is possible that there may be some double counting of patient populations. This means that the CIs may be slightly narrower than they would have been if all duplicated patients had been excluded. Many of the studies were at high risk of bias as they drew a comparison cohort from general population rates. The majority of studies did not adjust for cardiovascular risk factors. Furthermore, it was not possible to use reported results that were adjusted for important confounders as there was inconsistency in the confounders reported.

Despite including conference abstracts, there was some evidence of small study bias among studies reporting CHD and circulatory mortality for schizophrenia. These were studies with a higher risk of bias that used general population mortality rates to report SMRs. However, even after adjustment for possible unpublished studies, the risk of excess mortality remained elevated.

Although the literature searches found many papers measuring the association between SMI and the risk of cardiovascular mortality, searching one database for the additional search term "cause of death" revealed some additional studies. Thus, a small number of relevant studies may have been missed by failing to include this term in all database searches. However, any effect on the results is likely to be small due to the large number of included studies.

SMRs were calculated from available information if not reported directly (<u>S24 File</u>). If only whole numbers of expected deaths were available, this may have led to rounding errors. This is unlikely to have had much effect on the results.

Change over time. Temporal changes were assessed using decade of median year of follow-up. By comparing estimates from different studies from different time periods, we are performing an ecological analysis. Studies could have had outcomes spanning multiple decades but would only have been counted against one. This methodology may have reduced the apparent difference in relative risk between decades highlighted in the review. More insight on temporal changes may be gained by considering large, long-term cohort studies, and the use of individual patient data.

SMI definitions. There was some inconsistency in definitions of SMI diagnosis, which could have accounted for some of the heterogeneity between studies. Most studies used DSM or ICD coding to define SMI, but some defined SMI less robustly as confirmed by psychiatrist or otherwise recorded in notes [77].

Around half the included studies reporting CVD outcomes for schizophrenia grouped schizoaffective disorder together with schizophrenia. Another study grouped schizoaffective disorder with BD [97]. Schizophrenia causes more significant disability than BD [98]; schizoaffective disorder has symptoms of both conditions, so its inclusion could have reduced the average severity of schizophrenia exposure or increased it for BD [99].

Some definitions of schizophrenia included schizotypal disorder [89,100–106] (a type of nonpsychotic personality disorder below the clinical threshold for schizophrenia) [36]. Others did not specify either the diagnostic system or the codes used [45,107–113], so severity was not able to be determined. The severity of schizophrenia cases could also have been reduced by the exclusion of schizophrenia with substance use disorder [101].

Although studies that included psychotic depression were eligible for inclusion, other forms of major depressive disorder were not, due to the difficulty in distinguishing major depression from other types and severity of depression.

Studies where all participants were treated with lithium included those diagnosed with BD and schizoaffective disorder. However, unipolar depression and other SMI diagnoses that did not meet inclusion criteria were also present [97,114,115]. It was not possible to exclude these people, so some misclassified patients were included, although numbers were small. Categorisation of SMI in other studies may also be subject to misclassification due to variation in psychiatric diagnostic labelling [116].

CVD ascertainment. Most studies did not demonstrate adequately that CVD was not present at the start of the study. Although studies were selected such that SMI was diagnosed before cardiovascular death, presence of diagnosed or undiagnosed CVD prior to death was unknown for most studies. In incidence studies, although it is likely that SMI diagnosis was made at a younger age than incidence of CVD, it was not possible to be confident that CVD did occur after the start of SMI diagnosis.

Age. Studies were included regardless of the participants' age at cardiovascular outcome, which was not known for most studies. However, larger associations between SMI and CVD were noted among mortality studies that did report relative risks for younger age groups. This could be due to a healthy survivor effect: Those with less SMI are more likely to survive to older ages.

Confounding factors. Not many mortality studies adjusted for confounding factors, other than age and sex, so exploration of the effects of important covariates, such as smoking, ethnicity, deprivation, and other cardiovascular risk factors was not possible. However, some studies noted that antipsychotics were associated with higher risks of cardiovascular mortality [87,117,118]. More research is needed to assess the effects of confounders on CVD mortality.

Most of the incidence studies did not adjust for important confounders with inconsistency in the confounders reported for different studies. Using results that were adjusted for confounders made little difference to the results, but it is likely that adjusting for factors such as smoking, comorbidities, obesity, or antipsychotic use may explain the increased risks.

Setting. Included studies were generally representative of persons with SMI. Those that included only inpatients were likely to have missed less severe cases, potentially overestimating risks, although no significant heterogeneity due to setting was detected. Community settings were underrepresented in the review: Many of the recruited exposed populations were inpatients. Consequently, the review includes an overrepresentation of participants with more severe symptoms of exposure and outcome. A lack of patients with relatively mild symptoms from primary care settings may explain some of the apparent underdiagnosis of CHD seen in the review.

Risk of bias. There was a risk of selection or information bias affecting many of the included studies. The risk of CVA diagnosis and total CVD events found for all studies remained elevated among the high-quality studies. However, risk of CHD diagnosis was elevated only for the high-quality studies.

Reporting by sex. Results for all persons were included where possible, but for some studies, data were only available separately for males and females. In these cases, results from each sex were included separately in the meta-analyses. However, results for males and females within the same study are not independent, so the random effects assumption of the metaanalysis could be violated. In practice, this is unlikely to have much effect on the conclusions.

Conclusions

This systematic review and meta-analysis found that people with SMI are at increased risk of mortality from CVD compared with the general population and that the risk may be increasing over time. It also found that the risk of diagnosis with stroke, heart failure, and major cardiovascular events was elevated for people with SMI compared with control participants.

It was not possible to explore the effects of confounders, such as smoking and obesity, so at least some of the excess risk may be attributable to these residual factors. There was considerable heterogeneity between studies, partly explained by decade of results, world region, and age group. Prediction intervals included unity for some combinations of SMI and CVD mortality; thus, a future study may not find an increased risk of CVD mortality for people with SMI. However, this review included results from 82 studies, only 7 of which reported any estimated risks smaller than one. Any future study is unlikely to materially change the review findings.

Deaths from CVD make a large contribution to the inequalities in life expectancy experienced by those with SMI [6]. This result highlights the need for targeted interventions to improve the CVD health of people with SMI. A better mechanistic understanding of the reasons for the higher mortality rates would enable identification of where inequalities could be reduced. More research is needed to explore the role of smoking, obesity, antipsychotic medication, and access as well as adherence to CVD treatments for the excess cardiovascular mortality.

More research is needed to understand the reasons for the higher morbidity risk and to assess why it may have been worsening in recent decades. Such research should include the investigation of the relationship to cardiovascular risk factors, such as obesity, diabetes, blood pressure, smoking, and antipsychotics, and the role of access and adherence to treatments for these risk factors. Cohort studies accounting for these, and other important confounders, could help further investigate these factors.

Author contributions

AML conducted the literature searches, carried out the analysis, and drafted the review. TM, HMP, and RR critically appraised and provided guidance during the development of the protocol and the review process. TM and HMP edited and redrafted the manuscript. AML, EP, HMP, TM, MO, MR, NT, TSA, GM, AL, and KA screened titles and abstracts or reviewed full texts. AML and EP conducted data extraction and risk of bias assessment. MJP provided advice on statistical analysis. CUC and MS contributed data from a previous review and the data extraction form. All authors read, edited, and approved the final version of the final manuscript.

Supporting information

S1 File. PRISMA Checklist. PRISMA, Preferred Reporting Items for Systematic Review and Meta-Analyses. (DOCX)

S2 File. PRISMA Abstract Checklist. PRISMA, Preferred Reporting Items for Systematic Review and Meta-Analyses.

(DOCX)

S3 File. Deviations from protocol. (DOCX)

S4 File. Inclusion and exclusion criteria. (DOCX)

S5 File. Search strategies by database. (DOCX)

S6 File. Data extraction form. (DOCX)

S7 File. Risk of bias forms for mortality and incidence studies. Table A: Risk of bias form for cohort studies. Table B: Risk of bias form for case–control studies. (DOCX)

S8 File. Decision rules for overlapping studies. (DOCX)

S9 File. List of included mortality studies. (DOCX)

S10 File. List of included incidence studies. (DOCX)

S11 File. List of excluded studies. (DOCX)

S12 File. Characteristics of included mortality studies. (DOCX)

S13 File. Characteristics of included incidence studies. (DOCX)

S14 File. Forest plots reporting cardiovascular mortality outcomes, across decades. Fig A: Schizophrenia, mortality from CVA, SMRs. Fig B: Schizophrenia, mortality from CHD, SMRs. Fig C: Schizophrenia, mortality from all circulatory disease, SMRs. Fig D: BD, mortality from CVA, SMRs. Fig E: BD, mortality from CHD, SMRs. Fig F: BD, mortality from all circulatory disease, SMRs. Fig G: Schizophrenia, mortality from CVA, HRs, rate ratios and ORs. Fig H: Schizophrenia, mortality from CHD, HRs, rate ratios and ORs. Fig I: Schizophrenia, mortality from heart failure, HRs, rate ratios and ORs. Fig K: BD, mortality from CHD, HRs, rate ratios and ORs. Fig M: BD, mortality from heart failure, HRs, rate ratios and ORs. Fig N: Mixed SMI, mortality from CVDs, HRs, rate ratios, ORs, SMRs. BD, bipolar disorder; CHD, coronary heart disease; CVA, cerebrovascular accident; HR, hazard ratio; OR, odds ratio; SMR, standardised mortality ratio. (PDF)

S15 File. Forest plots reporting cardiovascular incidence outcomes, across decades. Fig A: Schizophrenia, incidence of CVA, HRs, rate ratios and ORs. Fig B: Schizophrenia, incidence of CHD, HRs, rate ratios and ORs. Fig C: Schizophrenia, incidence of major cardiovascular events, HRs, rate ratios and ORs. Fig D: Schizophrenia, incidence of heart failure, HRs, rate

ratios and odds ratios. Fig E: BD, incidence of CVA, HRs, rate ratios, and ORs. Fig F: BD, incidence of CHD, HRs, rate ratios and ORs. Fig G: BD, incidence of major cardiovascular events, HRs, rate ratios and ORs. Fig H: BD, incidence of heart failure, HRs, rate ratios and ORs. BD, bipolar disorder; CHD, coronary heart disease; CVA, cerebrovascular accident; HR, hazard ratio; OR, odds ratio.

(PDF)

S16 File. Risk of bias assessment for included mortality studies. Table A: Risk of bias assessment of studies reporting cardiovascular mortality outcomes, cohort studies. (DOCX)

S17 File. Risk of bias assessment for included incidence studies. Table A: Risk of bias assessment of studies reporting cardiovascular incidence outcomes, cohort studies. Table B: Risk of bias assessment of studies reporting cardiovascular incidence outcomes, case-control studies. (DOCX)

S18 File. Meta-analysis of CVD incidence outcomes. Fig A: Forest plots showing relative risk of CVD incidence in those with versus without schizophrenia, studies included in meta-analysis. Fig B: Forest plots showing relative risk of CVD incidence in those with versus without BD, studies included in meta-analysis. BD, bipolar disorder; CVD, cardiovascular disease. (PDF)

S19 File. Analysis of temporal trends. Fig A: Forest plot showing pooled estimates of cardiovascular mortality ratios by SMI, outcome and decade. Fig B: Forest plot showing pooled estimates of cardiovascular incidence ratios by SMI, outcome and decade. Table A: Meta-analysis: pooled results by SMI and cardiovascular mortality. Table B: Meta-analysis: pooled results by SMI and cardiovascular incidence. Table C: Results of meta-regressions: estimates of increased effect size relative to the reference category by median 5-year calendar period of outcome, exponentiated regression coefficients (95% CIs,) *p*-values. Fig C: Trend in risk of CVD mortality for SMI compared with controls, by median 5-year calendar period of outcome. Fig D: Trend in risk of CVD incidence for SMI compared with controls, by median 5-year calendar period of outcome. CI, confidence interval; CVD, cardiovascular disease; SMI, severe mental illness.

(PDF)

S20 File. Subgroup analysis. Fig A: Forest plots showing studies reporting CVD mortality risk for SMI versus controls for (a) age groups \leq 60; (b) older age groups or all ages. Fig B: Forest plots showing relative risk of CVD mortality in those with versus without schizophrenia, studies included in meta-analysis for (a) males; (b) females. Fig C: Forest plots showing relative risk of CVD mortality in those with versus without BD, studies included in meta-analysis for (a) males; (b) females. Fig D: Forest plots showing relative risk of CVD mortality in those with versus without BD, studies included in meta-analysis for (a) males; (b) females. Fig D: Forest plots showing relative risk of CVD mortality in those with versus without mixed SMI, studies included in meta-analysis for (a) males; (b) females. Fig E: Forest plots of CVD incidence for (a) studies with low risk of bias, (b) all studies included in meta-analysis, schizophrenia and BD. BD, bipolar disorder; CVD, cardiovascular disease; SMI, severe mental illness.

(PDF)

S21 File. Sensitivity analysis. Table A: Sensitivity analyses showing effect of replacing included with excluded results, studies reporting mortality outcomes. Table B: Sensitivity analyses showing effect of replacing included with excluded results, studies reporting incidence outcomes. Table C: Pooled relative risks of cardiovascular incidence by SMI and CVD outcome, comparison of minimally adjusted, and fully adjusted models reported by included

studies. CVD, cardiovascular disease; SMI, severe mental illness. (DOCX)

S22 File. Assessment of publication bias: mortality studies. Fig A: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for schizophrenia compared with controls, CVA. Fig B: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for schizophrenia compared with controls, CHD. Fig C: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for schizophrenia compared with controls, all circulatory disease. Fig D: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for BD compared with controls, CVA. Fig E: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for BD compared with controls, CHD. Fig F: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD mortality for BD compared with controls, all circulatory disease. Table A: Results of Egger tests for publication bias, mortality outcomes. Fig G: Funnel plot of schizophrenia and CHD mortality including 5 unpublished studies estimated from trim and fill. Fig H: Funnel plot of schizophrenia and all circulatory disease mortality including 2 unpublished studies estimated from trim and fill. BD, bipolar disorder; CHD, coronary heart disease; CVA, cerebrovascular accident; CVD, cardiovascular disease. (PDF)

S23 File. Assessment of publication bias: incidence studies. Fig A: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia compared with controls, CVA. Fig B: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia compared with controls, CHD. Fig C: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia compared with controls, major cardiovascular events. Fig D: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia compared with controls, heart failure. Fig E: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for BD compared with controls, CVA. Fig F: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for BD compared with controls, CHD. Fig G: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for BD compared with controls, major cardiovascular events. Fig H: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for BD compared with controls, heart failure. Fig I: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia compared with controls, all CVD. Fig J: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for BD compared with controls, all CVD. Fig K: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia and BD compared with controls, CVA. Fig L: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia and BD compared with controls, CHD. Fig M: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia and BD compared with controls, major cardiovascular events. Fig N: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia and BD compared with controls, heart failure. Fig O: Funnel plots for visual assessment of publication bias for studies reporting relative risk of CVD incidence for schizophrenia and BD compared with controls, all CVD. Table A: Results of Egger tests for publication bias, incidence outcomes. BD, bipolar disorder; CHD, coronary heart

disease; CVA, cerebrovascular accident; CVD, cardiovascular disease. (PDF)

S24 File. Example calculations for estimating unreported cases. (DOCX)

Acknowledgments

The authors would like to thank Danai Bem for screening titles and abstracts and providing guidance and critical appraisal during the development of the protocol and the review process and Arwa Abdel-Aal for help with piloting the title and abstract screening used in this systematic review.

Disclaimers

The views expressed in this publication are those of the authors and not necessarily those of the NIHR, NHS, the UK Department of Health and Social Care or other funding institutions.

Author Contributions

Conceptualization: Amanda M Lambert, Helen M Parretti, Ronan Ryan, Tom Marshall.

Formal analysis: Amanda M Lambert, Emma Pearce, Malcolm J Price.

Funding acquisition: Tom Marshall.

- **Investigation:** Amanda M Lambert, Helen M Parretti, Emma Pearce, Mark Riley, Natalie Tyldesley-Marshall, Tuba Saygın Avşar, Gemma Matthewman, Alexandra Lee, Khaled Ahmed, Maria Lisa Odland, Tom Marshall.
- **Methodology:** Amanda M Lambert, Helen M Parretti, Emma Pearce, Malcolm J Price, Marco Solmi, Tom Marshall.

Project administration: Amanda M Lambert.

Resources: Amanda M Lambert, Christoph U. Correll, Marco Solmi.

Supervision: Helen M Parretti, Malcolm J Price, Ronan Ryan, Tom Marshall.

Visualization: Amanda M Lambert.

Writing - original draft: Amanda M Lambert, Helen M Parretti, Tom Marshall.

Writing – review & editing: Amanda M Lambert, Helen M Parretti, Emma Pearce, Malcolm J Price, Mark Riley, Ronan Ryan, Natalie Tyldesley-Marshall, Tuba Saygın Avşar, Gemma Matthewman, Alexandra Lee, Khaled Ahmed, Maria Lisa Odland, Christoph U. Correll, Marco Solmi, Tom Marshall.

References

- 1. Walker E, McGee RE, Druss BG. Mortality in mental disorders and global disease burden implications: A systematic review and meta-analysis. JAMA Psychiat. 2015; 72(4):334–41. https://doi.org/10.1001/ jamapsychiatry.2014.2502 PMID: 25671328
- Hjorthøj C, Stürup AE, McGrath JJ, Nordentoft M. Years of potential life lost and life expectancy in schizophrenia: a systematic review and meta-analysis. Lancet Psychiatry. 2017; 4(4):295–301. https://doi.org/10.1016/S2215-0366(17)30078-0 PMID: 28237639
- Pan Y-J, Yeh L-L, Chan H-Y, Chang C-K. Excess mortality and shortened life expectancy in people with major mental illnesses in Taiwan. Epidemiol Psychiatr Sci. 2020; 29:e156. Epub 2020/08/14. https://doi.org/10.1017/S2045796020000694 PMID: 32792024

- Kessing LV, Vradi E, Andersen PK. Life expectancy in bipolar disorder. Bipolar Disord. 2015; 17 (5):543–8. https://doi.org/10.1111/bdi.12296 PMID: 25846854
- Correll CU, Solmi M, Veronese N, Bortolato B, Rosson S, Santonastaso P, et al. Prevalence, incidence and mortality from cardiovascular disease in patients with pooled and specific severe mental illness: A large-scale meta-analysis of 3,211,768 patients and 113,383,368 controls. World Psychiatry. 2017; 16(2):163–80. https://doi.org/10.1002/wps.20420 PMID: 28498599
- De Hert M, Correll CU, Bobes J, Cetkovich-Bakmas M, Cohen D, Asai I, et al. Physical illness in patients with severe mental disorders. I. Prevalence, impact of medications and disparities in health care. World Psychiatry. 2011;10. https://doi.org/10.1002/j.2051-5545.2011.tb00014.x PMID: 21379357
- De Hert M, Cohen D, Bobes J, Cetkovich-Bakmas M, Leucht S, Ndetei DM, et al. Physical illness in patients with severe mental disorders. II. Barriers to care, monitoring and treatment guidelines, plus recommendations at the system and individual level. World Psychiatry. 2011;10. https://doi.org/10. 1002/j.2051-5545.2011.tb00036.x PMID: 21633691
- Mitchell AJ, Lawrence D. Revascularisation and mortality rates following acute coronary syndromes in people with severe mental illness: comparative meta-analysis. BrJ Psychiatry. 2011; 198(6):434–41. Epub 2011/06/02. https://doi.org/10.1192/bjp.bp.109.076950 PMID: 21628705.
- Mitchell AJ, Malone D, Doebbeling CC. Quality of medical care for people with and without comorbid mental illness and substance misuse: systematic review of comparative studies. Br J Psychiatry. 2009; 194(6):491–9. https://doi.org/10.1192/bjp.bp.107.045732 PMID: 19478286
- Mitchell A, Lord O. Review: do deficits in cardiac care influence high mortality rates in schizophrenia? A systematic review and pooled analysis. J Psychopharmacol. 2010; 24(Suppl. 4):69–80. https://doi. org/10.1177/1359786810382056 PMID: 20923922
- 11. Mitchell A, Lord O, Malone D. Differences in the prescribing of medication for physical disorders in individuals with v. without mental illness: meta-analysis. Br J Psychiatry. 2012; 201(6):435–43. <u>https://doi.org/10.1192/bjp.bp.111.094532</u> PMID: 23209089 doi:
- Heiberg IH, Jacobsen BK, Balteskard L, Bramness JG, Naess O, Ystrom E, et al. Undiagnosed cardiovascular disease prior to cardiovascular death in individuals with severe mental illness. Acta Psychiatr Scand. 2019; 139(6):558–71. https://doi.org/10.1111/acps.13017 PMID: 30844079
- 13. Institute for Health Metrics and Evaluation (IHME). GHDX Results Tool: Cardiovascular deaths. Seattle, WA: IHME, University of Washington; 2021 [2 Jun 2021]. Available from: http://ghdx.healthdata. org/gbd-results-tool?params=gbd-api-2019-permalink/058e5013f3d86db8c78488e5b2dd2df6.
- McGrath J, Saha S, Chant D, Welham J. Schizophrenia: A Concise Overview of Incidence, Prevalence, and Mortality. Epidemiol Rev. 2008; 30(1):67–76. https://doi.org/10.1093/epirev/mxn001 PMID: 18480098
- Jablensky A. Epidemiology of schizophrenia: the global burden of disease and disability. Eur Arch Psychiatry Clin Neurosci. 2000; 250(6):274–85. https://doi.org/10.1007/s004060070002 PMID: 11153962
- Institute for Health Metrics and Evaluation (IHME). GHDX Results Tool: SMI DALYs. Seattle, WA: IHME, University of Washington; 2021 [2 Jun 2021]. Available from: http://ghdx.healthdata.org/gbdresults-tool?params=gbd-api-2019-permalink/f1bf332b48db9a88a89cd75cca013fd2.
- Amini M, Zayeri F, Salehi M. Trend analysis of cardiovascular disease mortality, incidence, and mortality-to-incidence ratio: results from global burden of disease study 2017. BMC Public Health. 2021; 21 (1):401. https://doi.org/10.1186/s12889-021-10429-0 PMID: 33632204
- Institute for Health Metrics and Evaluation (IHME). GHDX. Results Tool: CVD incidence. Seattle, WA: IHME, University of Washington; 2021 [2 Jun 2021]. Available from: http://ghdx.healthdata.org/gbdresults-tool?params=gbd-api-2019-permalink/55968ff0ff22563ae352c89851c9400d.
- Joseph P, Leong D, McKee M, Anand SS, Schwalm J-D, Teo K, et al. Reducing the Global Burden of Cardiovascular Disease, Part 1. Circ Res. 2017; 121(6):677–94. <u>https://doi.org/10.1161/</u> CIRCRESAHA.117.308903 PMID: 28860318
- 20. O'Brien EC, Rose KM, Suchindran CM, Sturmer T, Chang PP, Alonso A, et al. Temporal Trends in Medical Therapies for ST- and Non-ST Elevation Myocardial Infarction: (from the Atherosclerosis Risk in Communities [ARIC] Surveillance Study). Am J Cardiol. 2013; 111(3):305–11. https://doi.org/10. 1016/j.amjcard.2012.09.032 PMID: 23168284
- Hayes JF, Marston L, Walters K, King MB, Osborn DP, Osborn DPJ. Mortality gap for people with bipolar disorder and schizophrenia: UK-based cohort study 2000–2014. Br J Psychiatry. 2017; 211 (3):175–81. https://doi.org/10.1192/bjp.bp.117.202606 PMID: 286844037.
- 22. Prieto ML, Cuéllar-Barboza AB, Bobo WV, Roger VL, Bellivier F, Leboyer M, et al. Risk of myocardial infarction and stroke in bipolar disorder: a systematic review and exploratory meta-analysis. Acta Psychiatr Scand. 2014; 130(5):342–53. https://doi.org/10.1111/acps.12293 PMID: 24850482

- Li M, Fan YL, Tang ZY, Cheng XS. Schizophrenia and risk of stroke: A meta-analysis of cohort studies. Int J Cardiol. 2014; 173(3):588–90. https://doi.org/10.1016/j.ijcard.2014.03.101 PMID: 24698235
- 24. Saha S, Chant D, McGrath J. A systematic review of mortality in schizophrenia: Is the differential mortality gap worsening over time? Arch Gen Psychiatry. 2007; 64(10):1123–31. https://doi.org/10.1001/ archpsyc.64.10.1123 PMID: 17909124
- Fan Z, Wu Y, Shen J, Ji T, Zhan R. Schizophrenia and the risk of cardiovascular diseases: A metaanalysis of thirteen cohort studies. J Psychiatr Res. 2013; 47(11):1549–56. https://doi.org/10.1016/j. jpsychires.2013.07.011 PMID: 23953755
- Roshanaei-Moghaddam B, Katon W. Premature Mortality From General Medical Illnesses Among Persons With Bipolar Disorder: A Review. Psychiatr Serv. 2009; 60(2):147–56. <u>https://doi.org/10.1176/ps.2009.60.2.147</u> PMID: 19176408
- Hayes J, Miles J, Walters K, King M, Osborn D. A systematic review and meta-analysis of premature mortality in bipolar affective disorder. Acta Psychiatr Scand. 2015; 131(6):417–25. <u>https://doi.org/10.1111/acps.12408</u> PMID: 25735195.
- Brown S. Excess mortality of schizophrenia. A meta-analysis Br J Psychiatry. 1997; 171:502–8. https://doi.org/10.1192/bjp.171.6.502 PMID: 9519087.
- 29. Harris C, Barraclough B. Excess mortality of mental disorder. Br J Psychiatry. 1998; 173(1):11–53. Epub 2018/01/03. https://doi.org/10.1192/bjp.173.1.11 PMID: 9850203
- Lumme S, Pirkola S, Manderbacka K, Keskimaki I. Excess Mortality in Patients with Severe Mental Disorders in 1996–2010 in Finland. PLoS ONE. 2016; 11(3):e0152223. <u>https://doi.org/10.1371/journal.pone.0152223</u> PMID: 27010534
- **31.** Page MJ, McKenzie JE, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. The PRISMA 2020 statement: An updated guideline for reporting systematic reviews. PLoS Med. 2021; 18(3):e1003583. https://doi.org/10.1371/journal.pmed.1003583 PMID: 33780438
- Stroup DF, Berlin JA, Morton SC, Olkin I, Williamson GD, Rennie D, et al. Meta-analysis of Observational Studies in EpidemiologyA Proposal for Reporting. JAMA. 2000; 283(15):2008–12. <u>https://doi.org/10.1001/jama.283.15.2008 PMID</u>: 10789670
- McManus S, Meltzer H, Brugha T.S., Bebbington P.E., Jenkins R. Adult psychiatric morbidity in England, 2007: Results of a household survey. London: The NHS Information Centre for health and social care, 2009.
- Kørner A, Lopez AG, Lauritzen L, Andersen PK, Kessing LV. Acute and transient psychosis in old age and the subsequent risk of dementia: A nationwide register-based study. Geriatr Gerontol Int. 2009; 9 (1):62–8. https://doi.org/10.1111/j.1447-0594.2009.00505.x PMID: 19260981
- 35. International Monetary Fund. World Economic Outlook: Too Slow for Too Long. Washington: 2016.
- **36.** American Psychiatric Association. Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition. Arlington, VA: American Psychiatric Association; 2013.
- World Health Organisation. ICD-10 classification of mental and behavioural disorders: clinical descriptions and diagnostic guidelines. Geneva: World Health Organisation; 1992.
- McGuffin P, Farmer A, Harvey I. A Polydiagnostic Application of Operational Criteria in Studies of Psychotic Illness: Development and Reliability of the OPCRIT System. Arch Gen Psychiatry. 1991; 48 (8):764–70. https://doi.org/10.1001/archpsyc.1991.01810320088015 PMID: 1883262
- Chisholm J. The Read clinical classification. BMJ. 1990; 300(6732):1092. https://doi.org/10.1136/bmj. 300.6732.1092 PMID: 2344534.
- Office for National Statistics. Deaths registered by cause, gender and age 2019 [June 1, 2020]. Available from: https://www.nomisweb.co.uk/home/release_group.asp?g=23.
- 41. NHS Digital. Hospital Episode Statistics 2018/19 2019 [Jun 1, 2020]. Available from: https://digital. nhs.uk/data-and-information/publications/statistical/hospital-admitted-patient-care-activity/2018-19.
- Termorshuizen F, Wierdsma AI, Smeets HM, Visser E, Drukker M, Nijman H, et al. Cause-Specific Mortality Among Patients With Psychosis: Disentangling the Effects of Age and Illness Duration. Psychosomatics. 2013; 54(6):536–45. https://doi.org/10.1016/j.psym.2013.05.011 PMID: 24012288.
- Ösby U, Westman J, Hällgren J, Gissler M. Mortality trends in cardiovascular causes in schizophrenia, bipolar and unipolar mood disorder in Sweden 1987–2010. Eur J Public Health. 2016; 26(5):867–71. Epub 2016/01/10. https://doi.org/10.1093/eurpub/ckv245 PMID: 26748100; PubMed Central PMCID: PMC5054269.
- 44. Laursen TM, Nordentoft M. Heart disease treatment and mortality in schizophrenia and bipolar disorder-Changes in the Danish population between 1994 and 2006. J Psychiatr Res. 2011; 45(1):29–35. https://doi.org/10.1016/j.jpsychires.2010.04.027 PMID: 20546788

- Tsuang MT, Woolson RF, Fleming JA. Causes of death in schizophrenia and manic-depression. Br J Psychiatry. 1980; 136:239–42. https://doi.org/10.1192/bjp.136.3.239 PMID: 7388226.
- Kirkwood BR, Sterne JAC. Essential Medical Statistics. Malden, Massachusetts: Blackwell Science; 2003.
- Wells GAS, B.; O'Connell, D.; Peterson, J.; Welch, V.; Losos, M.; Tugwell, P. The Newcastle-Ottawa Scale (NOS) for assessing the quality of nonrandomized studies in meta-analyses [31 Oct 2019]. Available from: http://www.ohri.ca/programs/clinical_epidemiology/oxford.asp.
- Critical Appraisal Skills Programme. CASP cohort study checklist 2018 [March 19, 2020]. Available from: https://casp-uk.net/wp-content/uploads/2018/01/CASP-Cohort-Study-Checklist_2018.pdf.
- DerSimonian R, Laird N. Meta-analysis in clinical trials. Control Clin Trials. 1986; 7(3):177–88. <u>https://</u> doi.org/10.1016/0197-2456(86)90046-2 PMID: 3802833
- Higgins JPT, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. BMJ. 2003; 327(7414):557–60. https://doi.org/10.1136/bmj.327.7414.557 PMID: 12958120
- Page MJ, Higgins JPT, Sterne JAC. Chapter 13: Assessing risk of bias due to missing results in a synthesis. 2019. In: Cochrane Handbook for Systematic Reviews of Interventions, Version 6.0. Cochrane, [cited June 12, 2020]. Available from: www.training.cochrane.org/handbook.
- Sterne JAC, Egger M, Moher D. Chapter 10: Addressing reporting biases. In: Higgins J, Green S, editors. Cochrane Handbook for Systematic Reviews of Interventions. Version 5.1.0 [updated March 2011] ed: The Cochrane Collaboration; 2011.
- **53.** Sterne JAC, Egger M, Davey Smith G. Investigating and dealing with publication and other biases. 2001. In: Systematic reviews in health care: meta-analysis in contex. BMJ Books.
- 54. Deeks JJ, Higgins JPT, Altman DG. Chapter 10: Analysing data and undertaking meta-analyses. 2019. In: Cochrane Handbook for Systematic Reviews of Interventions. Cochrane. Version 6.0. [cited June 12, 2020]. Available from: www.training.cochrane.org/handbook.
- 55. Harbord RM, Higgins JPT. Meta-Regression in Stata. Stata J. 2008; 8(4):493–519. https://doi.org/10. 1177/1536867X0800800403
- 56. Higgins JPT, López-López JA, Aloe AM. Chapter 7: Meta-Regression. 2020. In: Handbook of Meta-Analysis. Chapman and Hall/CRC.
- Bent-Ennakhil N, Cecile Perier M, Sobocki P, Gothefors D, Johansson G, Milea D, et al. Incidence of cardiovascular diseases and type-2-diabetes mellitus in patients with psychiatric disorders. Nord J Psychiatry. 2018; 72(7):455–61. https://doi.org/10.1080/08039488.2018.1463392 PMID: 30513230
- Brink M, Green A, Bojesen AB, Lamberti JS, Conwell Y, Andersen K. Excess medical comorbidity and mortality across the lifespan in schizophrenia.: A nationwide Danish register study. Schizophr Res. 2018. https://doi.org/10.1016/j.schres.2018.10.020 PMID: 30527270
- Crump C, Sundquist K, Sundquist J. Comorbidities and mortality in persons with schizophrenia: a Swedish national cohort study. Am J Psychiatry. 2013; 170:324–33. <u>https://doi.org/10.1176/appi.ajp.</u> 2012.12050599 PMID: 23318474.
- Crump C, Sundquist K, Winkleby MA, Sundquist J. Comorbidities and mortality in bipolar disorder: A swedish national cohort study. JAMA Psychiat. 2013; 70(9):931–9. https://doi.org/10.1001/ jamapsychiatry.2013.1394 PMID: 23863861
- Gale CR, Batty GD, Osborn DPJ, Tynelius P, Rasmussen F. Mental disorders across the adult life course and future coronary heart disease: evidence for general susceptibility. Circulation. 2014; 129 (2):186–93. https://doi.org/10.1161/CIRCULATIONAHA.113.002065 PMID: 241909591.
- Foroughi M, Medina Inojosa JR, Lopez-Jimenez F, Saeidifard F, Suarez L, Stokin GB, et al. Association of Bipolar Disorder With Major Adverse Cardiovascular Events: A Population-Based Historical Cohort Study. Psychosom Med. 2022; 84(1). https://doi.org/10.1097/PSY.00000000001017 PMID: 34611111
- Lin HC, Chen YH, Lee HC. Increased risk of acute myocardial infarction after acute episode of schizophrenia: 6 year follow-up study. Aust N Z J Psychiatry. 2010; 44(3):273–9. <u>https://doi.org/10.3109/</u> 00048670903487209 PMID: 20050720
- Prieto ML, Schenck LA, Kruse JL, Klaas JP, Chamberlain AM, Bobo WV, et al. Long-term risk of myocardial infarction and stroke in bipolar I disorder: A population-based Cohort Study. J Affect Disord. 2016; 194:120–7. https://doi.org/10.1016/j.jad.2016.01.015 PMID: 26820761
- Sundquist K, Li X. Alcohol abuse partly mediates the association between coronary heart disease and affective or psychotic disorders: a follow-up study in Sweden. Acta Psychiatr Scand. 2006; 113 (4):283–9. https://doi.org/10.1111/j.1600-0447.2006.00774.x PMID: 16638072
- 66. Tsai KY, Lee CC, Chou YM, Su CY, Chou FH, Tsai K-Y, et al. The incidence and relative risk of stroke in patients with schizophrenia: a five-year follow-up study. Schizophr Res. 2012; 138(1):41–7. <u>https://</u> doi.org/10.1016/j.schres.2012.02.013 PMID: 223867341.

- 67. Wu HC, Chou FHC, Tsai KY, Su CY, Shen SP, Chung TC. The Incidence and Relative Risk of Stroke among Patients with Bipolar Disorder: A Seven-Year Follow-Up Study. PLoS ONE. 2013; 8(8): e73037. https://doi.org/10.1371/journal.pone.0073037 PMID: 24023667
- Yu-Chuan C, Ya-Mei B, Tung-Ping S, Tzeng-Ji C, Mu-Hong C, Chiu Y-C, et al. Ischemic Stroke in Young Adults and Preexisting Psychiatric Disorders: A Nationwide Case-Control Study. Medicine. 2015; 94(38):1–6. https://doi.org/10.1097/MD.000000000001520 PMID: 110556733.
- 69. Munk-Jorgensen P, Mors O, Mortensen PB, Ewald H. The schizophrenic patient in the somatic hospital. Acta Psychiatr Scand. 2000; 102:96–9. https://doi.org/10.1034/j.1600-0447.2000.00019.x PMID: WOS:000166790600020
- McDermott S, Moran R, Platt T, Isaac T, Wood H, Dasari S. Heart disease, schizophrenia, and affective psychoses: epidemiology of risk in primary care. Community Ment Health J. 2005; 41(6):747–55. https://doi.org/10.1007/s10597-005-6431-6 PMID: 163285876.
- Westman J, Hällgren J, Wahlbeck K, Erlinge D, Alfredsson L, Ösby U. Cardiovascular mortality in bipolar disorder: a population-based cohort study in Sweden. BMJ Open. 2013; 3(4). <u>https://doi.org/10. 1136/bmjopen-2012-002373</u> PMID: 23604348
- 72. Rassidaki NC, Athitakis C, Goulis C, Erotocritou A, Collarou T. Illnesses of the psychosomatic continuum and coronary heart disease. Int Ment Health Res Newsl. 1975; 17 (3):6.
- 73. Jensen S, Torp-Pedersen CTP, Skals RK, Aagaard J, Jensen SE. Increased prevalence of heart failure in patients with schizophrenia. Eur Heart J. 2017; 38(Suppl 1):912. <u>https://doi.org/10.1093/</u> eurheartj/ehx502.P4397
- Swain NR, Lim CCW, Levinson D, Fiestas F, de Girolamo G, Moskalewicz J, et al. Associations between DSM-IV mental disorders and subsequent non-fatal, self-reported stroke. J Psychosom Res. 2015; 79(2):130–6. https://doi.org/10.1016/j.jpsychores.2015.05.008 PMID: 260940100.
- 75. Nichols M, Townsend N, Scarborough P, Rayner M. Trends in age-specific coronary heart disease mortality in the European Union over three decades: 1980–2009. Eur Heart J. 2013; 34(39):3017–27. https://doi.org/10.1093/eurhearti/eht159 PMID: 23801825
- 76. Ösby U, Correia N, Brandt L, Ekbom A, Sparen P. Time trends in schizophrenia mortality in Stockholm County, Sweden: Cohort study. BMJ. 2000; 321(7259):483–4. https://doi.org/10.1136/bmj.321.7259. 483 PMID: 10948028
- Brown S, Kim M, Mitchell C, Inskip H. Twenty-five year mortality of a community cohort with schizophrenia. Br J Psychiatry. 2010; 196(2):116–21. <u>https://doi.org/10.1192/bjp.bp.109.067512</u> PMID: 20118455.
- Szatkowski L, McNeill A. Diverging trends in smoking behaviors according to mental health status. Nicotine Tob Res. 2015; 17(3):356–60. Epub 2014/09/01. <u>https://doi.org/10.1093/ntr/ntu173</u> PMID: 25180078.
- 79. Cook BL, Wayne GF, Kafali EN, Liu Z, Shu C, Flores M. Trends in Smoking Among Adults With Mental Illness and Association Between Mental Health Treatment and Smoking Cessation. JAMA. 2014; 311 (2):172–82. https://doi.org/10.1001/jama.2013.284985 PMID: 24399556
- National Institute for Health and Care Excellence. Psychosis and schizophrenia in adults: treatment and management. February 2014 [13 May 2017]. Available from: <u>https://www.nice.org.uk/guidance/ cg178</u>.
- Divac N, Prostran M, Jakovcevski I, Cerovac N. Second-Generation Antipsychotics and Extrapyramidal Adverse Effects. Biomed Res Int. 2014; 2014:656370. https://doi.org/10.1155/2014/656370 PMID: 24995318
- Hippius H. A historical perspective of clozapine. J Clin Psychiatry. 1999; 60(Suppl 12):22–3. Epub 1999/06/18. PMID: 10372606.
- Roberts R, Neasham A, Lambrinudi C, Khan A. A quantitative analysis of antipsychotic prescribing trends for the treatment of schizophrenia in England and Wales. JRSM Open. 2018; 9 (4):2054270418758570. Epub 2018/05/01. https://doi.org/10.1177/2054270418758570 PMID: 29707225; PubMed Central PMCID: PMC5912284.
- Rhee TG, Olfson M, Nierenberg AA, Wilkinson ST. 20-Year Trends in the Pharmacologic Treatment of Bipolar Disorder by Psychiatrists in Outpatient Care Settings. Am J Psychiatry. 2020; 177(8):706–15. Epub 2020/04/22. https://doi.org/10.1176/appi.ajp.2020.19091000 PMID: 32312111; PubMed Central PMCID: PMC7577523.
- Bohlken J, Bauer M, Kostev K. Drug treatment for patients with bipolar disorders in psychiatric practices in Germany in 2009 and 2018. Psychiatry Res. 2020; 289:112965. https://doi.org/10.1016/j. psychres.2020.112965 PMID: 32388174

- 86. Lyall LM, Penades N, Smith DJ. Changes in prescribing for bipolar disorder between 2009 and 2016: national-level data linkage study in Scotland. Br J Psychiatry. 2019; 215(1):415–21. Epub 2019/02/28. https://doi.org/10.1192/bjp.2019.16 PMID: 30816839
- **87.** Torniainen M, Mittendorfer-Rutz E, Tanskanen A, Bjorkenstam C, Suvisaari J, Alexanderson K, et al. Antipsychotic treatment and mortality in schizophrenia. Schizophr Bull. 2015; 41(3):656–63. <u>https://doi.org/10.1093/schbul/sbu164</u> PMID: 25422511; PubMed Central PMCID: PMC4393693.
- Lahti M, Tiihonen J, Wildgust H, Beary M, Hodgson R, Kajantie E, et al. Cardiovascular morbidity, mortality and pharmacotherapy in patients with schizophrenia. Psychol Med. 2012; 42(11):2275–85. https://doi.org/10.1017/S0033291712000396 PMID: 224055045.
- Manderbacka K, Arffman M, Sund R, Haukka J, Keskimaki I, Wahlbeck K. How does a history of psychiatric hospital care influence access to coronary care: a cohort study. BMJ Open. 2012; 2(2): e000831. Epub 2012/04/12. https://doi.org/10.1136/bmjopen-2012-000831 PMID: 22492387; PubMed Central PMCID: PMC3323812.
- Westman J, Eriksson SV, Gissler M, Hallgren J, Prieto ML, Bobo WV, et al. Increased cardiovascular mortality in people with schizophrenia: a 24-year national register study. Epidemiol Psychiatr Sci. 2018; 27(5):519–27. https://doi.org/10.1017/S2045796017000166 PMID: 28580898
- Jackson CA, Kerssens J, Fleetwood K, Smith DJ, Mercer SW, Wild SH. Incidence of ischaemic heart disease and stroke among people with psychiatric disorders: retrospective cohort study. Br J Psychiatry. 2020; 217(2):442–9. Epub 2019/11/22. https://doi.org/10.1192/bjp.2019.250 PMID: 31753047
- Ng M, Freeman MK, Fleming TD, Robinson M, Dwyer-Lindgren L, Thomson B, et al. Smoking Prevalence and Cigarette Consumption in 187 Countries, 1980–2012. JAMA. 2014; 311(2):183–92. https://doi.org/10.1001/jama.2013.284692 PMID: 24399557
- Bak M, Fransen A, Janssen J, van Os J, Drukker M. Almost All Antipsychotics Result in Weight Gain: A Meta-Analysis. PLoS ONE. 2014; 9(4):e94112. <u>https://doi.org/10.1371/journal.pone.0094112</u> PMID: 24763306
- Smith M, Hopkins D, Peveler RC, Holt RIG, Woodward M, Ismail K. First- v. second-generation antipsychotics and risk for diabetes in schizophrenia: Systematic review and meta-analysis. Br J Psychiatry. 2008; 192(6):406–11. Epub 2018/01/02. <u>https://doi.org/10.1192/bjp.bp.107.037184</u> PMID: 18515889
- Buhagiar K, Jabbar F. Association of First- vs. Second-Generation Antipsychotics with Lipid Abnormalities in Individuals with Severe Mental Illness: A Systematic Review and Meta-Analysis. Clin Drug Investig. 2019; 39(3):253–73. https://doi.org/10.1007/s40261-019-00751-2 PMID: 30675684
- Krumholz HM, Normand S-LT, Wang Y. Twenty-Year Trends in Outcomes for Older Adults With Acute Myocardial Infarction in the United States. JAMA Netw Open. 2019; 2(3):e191938–e. <u>https://doi.org/ 10.1001/jamanetworkopen.2019.1938</u> PMID: 30874787
- 97. Nilsson A. MORTALITY IN RECURRENT MOOD DISORDERS DURING PERIODS ON AND OFF LITHIUM—A COMPLETE POPULATION STUDY IN 362 PATIENTS. Pharmacopsychiatry. 1995; 28 (1):8–13. https://doi.org/10.1055/s-2007-979581 PMID: 7746844
- Salomon JA, Haagsma JA, Davis A, de Noordhout CM, Polinder S, Havelaar AH, et al. Disability weights for the Global Burden of Disease 2013 study. Lancet Glob Health. 2015; 3(11):e712–e23. https://doi.org/10.1016/S2214-109X(15)00069-8 PMID: 26475018
- 99. Lake CR, Hurwitz N. Schizoaffective disorder merges schizophrenia and bipolar disorders as one disease-there is no schizoaffective disorder. Curr Opin Psychiatry. 2007; 20(4). https://doi.org/10.1097/ YCO.0b013e3281a305ab PMID: 17551352
- 100. Grigoletti L, Perini G, Rossi A, Biggeri A, Barbui C, Tansella M, et al. Mortality and cause of death among psychiatric patients: a 20-year case-register study in an area with a community-based system of care. Psychol Med. 2009; 39(11):1875–84. https://doi.org/10.1017/S0033291709005790 PMID: 19379535.
- Heiberg IH, Jacobsen BK, Nesvag R, Bramness JG, Reichborn-Kjennerud T, Naess O, et al. Total and cause-specific standardized mortality ratios in patients with schizophrenia and/or substance use disorder. PLoS ONE. 2018; 13(8):e0202028. https://doi.org/10.1371/journal.pone.0202028 PMID: 30138449
- 102. Heilä H, Haukka J, Suvisaari J, Lönnqvist J. Mortality among patients with schizophrenia and reduced psychiatric hospital care. Psychol Med. 2005; 35(5):725–32. <u>https://doi.org/10.1017/</u> s0033291704004118 PMID: 159183493.
- 103. Hoye A, Jacobsen BK, Hansen V. Increasing mortality in schizophrenia: are women at particular risk? A follow-up of 1111 patients admitted during 1980–2006 in Northern Norway. Schizophr Res. 2011; 132(2–3):228–32. https://doi.org/10.1016/j.schres.2011.07.021 PMID: 21868200.

- 104. Kredentser MS, Martens PJ, Chochinov HM, Prior HJ. Cause and rate of death in people with schizophrenia across the lifespan: A population-based study in Manitoba. Canada J Clin Psychiatry. 2014; 75(2):154–60. https://doi.org/10.4088/JCP.13m08711 PMID: 24602251
- 105. Lesage A, Rochette L, Émond V, Pelletier É, St-Laurent D, Diallo FB, et al. A Surveillance System to Monitor Excess Mortality of People With Mental Illness in Canada. Can J Psychiatry. 2015; 60 (12):571–9. https://doi.org/10.1177/070674371506001208 PMID: 267208260.
- 106. Salazar-Fraile J, Gomez-Beneyto M, Perez-Hovos S, Hurtado-Navarro I. Mortality among psychiatric patients referred to the mental health services in Valencia. Soc Psychiatry Psychiatr Epidemiol. 1998; 33(5):224–9. https://doi.org/10.1007/s001270050047 PMID: 9604672
- 107. Angst F, Stassen HH, Clayton PJ, Angst J. Mortality of patients with mood disorders: follow-up over 34–38 years. J Affect Disord. 2002; 68 (2–3):167–81. https://doi.org/10.1016/s0165-0327(01)00377-9 PMID: 12063145.
- Brook OH. Mortality in the long-stay population of Dutch mental hospitals. Acta Psychiatr Scand. 1985; 71(6):626–35. Epub 1985/06/01. https://doi.org/10.1111/j.1600-0447.1985.tb02558.x PMID: 4024979.
- Buda M, Tsuang MT, Fleming JA. Causes of Death in DSM-III Schizophrenics and Other Psychotics (Atypical Group): A Comparison With the General Population. Arch Gen Psychiatry. 1988; 45(3):283– 5. https://doi.org/10.1001/archpsyc.1988.01800270101012 PMID: 3341883
- **110.** Giel R, Dijk S, van Weerden-Dijkstra JR. Mortality in the long-stay population of all Dutch mental hospitals. Acta Psychiatr Scand. 1978; 57(5):361–8. Epub 1978/05/01. <u>https://doi.org/10.1111/j.1600-0447</u>. 1978.tb06904.x PMID: 676779.
- Lawrence D, Hancock K, Kisely S. The gap in life expectancy from preventable physical illness in psychiatric patients in Western Australia: retrospective analysis of population based registers. BMJ. 2013; 346:f2539. https://doi.org/10.1136/bmj.f2539 PMID: 23694688.
- 112. Ösby U, Correia N, Brandt L, Ekbom A, Sparen P. Mortality and causes of death in schizophrenia in Stockholm county. Sweden Schizophr Res. 2000; 45(1–2):21–8. <u>https://doi.org/10.1016/s0920-9964</u> (99)00191-7 PMID: 10978869
- 113. Zilber N, Schufman N, Lerner Y. Mortality among psychiatric patients—the groups at risk. Acta Psychiatr Scand. 1989; 79(3):248–56. Epub 1989/03/01. https://doi.org/10.1111/j.1600-0447.1989. https://doi.org/10.1111/j.1600-0447. https://doi.org/10.1111/j.1600-0447. https://doi.org/10.1111/j.1600-0447. https://doi.org/10.1111/j.1600-0447. https://doi.org/10.1111/j.1600-0447. https://doi.org/10.11111/j.1600. <a href="https://doi.org/10.
- 114. Ahrens B, Müller-Oerlinghausen B, Schou M, Wolf T, Alda M, Grof E, et al. Excess cardiovascular and suicide mortality of affective disorders may be reduced by lithium prophylaxis. J Affect Disord. 1995; 33(2):67–75. https://doi.org/10.1016/0165-0327(94)00074-j PMID: 7759663
- 115. Brodersen A, Licht RW, Vestergaard P, Olesen AV, Mortensen PB. Sixteen-year mortality in patients with affective disorder commenced on lithium. Br J Psychiatry. 2000; 176:429–33. https://doi.org/10. 1192/bjp.176.5.429 PMID: 10912217
- Aboraya A, Rankin E, France C, El-Missiry A, John C. The Reliability of Psychiatric Diagnosis Revisited: The Clinician's Guide to Improve the Reliability of Psychiatric Diagnosis. Psychiatry (Edgmont). 2006; 3(1):41–50. PMID: 21103149.
- 117. Kelly DL, McMahon RP, Liu F, Love RC, Wehring HJ, Shim JC, et al. Cardiovascular disease mortality in patients with chronic schizophrenia treated with clozapine: A retrospective cohort study. J Clin Psychiatry. 2010; 71(3):304–11. https://doi.org/10.4088/JCP.08m04718yel PMID: 20079332
- 118. Osborn D, Levy G, Nazareth I, Petersen I, Islam A, King M. Relative risk of cardiovascular and cancer mortality in people with severe mental illness from the United Kingdom's general practice research database. Arch Gen Psychiatry. 2007; 64(2):242–9. <u>https://doi.org/10.1001/archpsyc.64.2.242</u> PMID: 17283292.